Published for SISSA by 🖄 Springer

RECEIVED: March 7, 2016 ACCEPTED: April 19, 2016 PUBLISHED: May 3, 2016

Superforms in six-dimensional superspace

Cesar Arias,^a William D. Linch III^b and Alexander K. Ridgway^c

^aDepartamento de Ciencias Físicas, Universidad Andres Bello, Republica 220, Santiago de Chile, Chile

^bCenter for String and Particle Theory, Department of Physics, University of Maryland at College Park, College Park, MD 20742-4111, U.S.A.

^cDepartment of Physics, California Institute of Technology, Pasadena, CA 91125, U.S.A. E-mail: ce.arias@uandresbello.edu, wdlinch3@gmail.com,

alecridgway@gmail.com

ABSTRACT: We investigate the complex of differential forms in curved, six-dimensional, N = (1,0) superspace. The superconformal group acts on this complex by super-Weyl transformations. An ambi-twistor-like representation of a second conformal group arises on a pure spinor subspace of the cotangent space. The *p*-forms are defined by super-Weylcovariant tensor fields on this pure spinor subspace. The on-shell dynamics of such fields is superconformal. We construct the superspace de Rham complex by successively obstructing the closure of forms. We then extend the analysis to composite forms obtained by wedging together forms of lower degree. Finally, we comment on applications to integration in curved superspace and propose a superspace formulation of the abelian limit of the nonabelian tensor hierarchy of N = (1, 0) superconformal models.

KEYWORDS: Superspaces, Differential and Algebraic Geometry, Extended Supersymmetry, Supergravity Models

ARXIV EPRINT: 1402.4823



Contents

1	Intr	roduct	1				
2	Closed superforms						
3	Conformal symmetry						
	3.1	7					
4	The super-differential complex						
	4.1	The c	11				
	4.2	The c	closed 2-form	12			
	4.3	The c	closed 3-form	13			
		4.3.1	Alternative formulation	15			
	4.4	The c	closed 4-form	15			
		4.4.1	Alternative formulation	17			
	4.5	The c	closed 5-form	17			
	4.6	The c	closed 6-form	19			
5	Cor	20					
	5.1	The c	21				
	5.2 The composite <i>p</i> -form with $p = 4, 5$, and 6			22			
		5.2.1	The composite 4-form	23			
		5.2.2	The composite 5-form	24			
		5.2.3	The composite 6-form	25			
6	Applications						
	6.1	Ectop	plasm	26			
	6.2 Abelian tensor hierarchy			27			
7	Cor	nclusio	on	29			
\mathbf{A}	Curved six-dimensional superspace						
в	The non-abelian tensor hierarchy						

JHEP05 (2016) 016

1 Introduction

The investigation of the structure of superforms is an important step toward understanding the geometry of the underlying superspace. Due to the interplay between the spinors and tensors in such spaces, this structure is non-trivial even in the flat case. In the flat, fourdimensional, N = 1 superspace, this is textbook material (cf. section 4.4 of *Superspace* [1] and the original ref. [2]).

In flat, four-dimensional, N = 2 harmonic superspace a systematic analysis was carried out by Biswas and Siegel [3]. In curved superspaces or in flat superspace of dimensions other than four, partial results may be found throughout the literature, with interesting developments in four and five dimensions being reported as recently as last year [4–8]. Systematic studies of the closely related integral invariants in various dimensions are being carried out by Howe and his collaborators (e.g. ref. [9–11].) Many of these results have been extended to three, five, and six dimensions even more recently by Kuzenko and collaborators (see e.g. ref. [12–16]). In ten dimensions, composite forms and their couplings to supergravity in superspace were used early on to construct effective actions for massless string states (e.g. ref. [17–19]). With the advent of covariant superstrings, a systematic analysis for the forms arising in the pure spinor superspace was performed by Berkovits and Howe [20] (see also [10]).

A special place in the hierarchy of superspaces is occupied by six-dimensional, N =(1,0) Minkowski space as it has the largest isometry group compatible with the existence of eight real supercharges. The auxiliary field problem (that there are strictly more fermionic auxiliary fields required for off-shell closure than bosonic ones [21]) is solved by retaining an $S^2 \simeq SU(2)_R/U(1)_R$ part of the *R*-symmetry group in the quotient construction [22, 23]. Closely related to this "harmonic" superspace is the "projective" superspace of [24–26]. The extension of this to curved, six-dimensional superspace was presented in [27] where certain super-Weyl-covariant field representations were defined and an action principle was proposed.¹ In this work, we continue the exploration of this superspace by analyzing the structure of the complex of differential forms. Motivated by its interpretation as the target space for covariant heterotic strings compactified on K3 [36], we introduce formal variables generating an algebra isomorphic to the graded exterior algebra in section 2. This algebra has a subalgebra corresponding to the projection of a bosonic spinor $s \mapsto \lambda \otimes v$ to the product of a pure spinor λ^{α} and iso-twistor variable v^i . We show that the differential forms are naturally defined on this subalgebra. In section 3, we exploit the existence of the pure spinor subspace of differential forms to define super-Weyl covariant tensor fields and construct an ambi-twistor-like representation of the six-dimensional conformal group that acts on these tensor fields. The resulting *superconformal* field representations are subsequently used in section 4 to derive the superspace analogue of the de Rham complex.² In section 5 we construct composite forms by wedging the various forms from section 4. In doing so, we are able to give a geometrical interpretation to certain multiplets constructed in [27]. Finally, we comment on some applications of the formalism in section 6. One of these is an attempt to reformulate the non-abelian tensor hierarchy [38-40], reviewed

¹This extension to six dimensions is based on previous work in five dimensions [28-31] and below [32-35].

²This complex was described recently as the Chevalley-Eilenberg complex of p-co-chains of the supersymmetry algebra with values in the space of superfields [8] (see also [7, 37]) where it was shown that the components and constraints have the interpretation of co-boundaries and co-homology, respectively, of the Chevalley-Eilenberg differential with values reduced to constant superfields.

in appendix B, in terms of a Chern-Simons theory on the ambi-twistor-like superspace of section 3. Additionally, since the Bianchi identity for the differential forms were solved in a curved background, this automatically determines the couplings of the abelian part of the tensor hierarchy to supergravity. Appendix A summarizes some results of reference [27] on curved, six-dimensional, N = (1, 0) superspace.

A note on notation. After careful consideration, we have decided to present our analysis (mostly) in an index-free notation because we are convinced that the benefits of doing so outweigh the risks. Firstly, this presentation most closely resembles our method of calculation and discovery and is useful in proofs. Secondly, the notation simplifies comparison to the covariant string theories and higher gauge theories to which it is closely related. Thirdly, we will need a representation of the conformal group defined in terms of these variables in order to build a certain differential complex of superfields in terms of which the *p*-form components are defined. Fourth, it is quite possible (and in fact happens in projective/harmonic and pure spinor theories) that some subset of these variables enters into the solution of constraints (such as those we will find) and/or the superspace measure when one attempts to define superinvariants by integration. Finally, the presentation of the resulting complex is more easily compared to the results on higher gauge theory existing in the literature. Conversely, using the more familiar, ordinary superspace notation would significantly complicate substantial parts of the presentation.

2 Closed superforms

In this section we write down the Bianchi identities for p-forms in six dimensions. We introduce some notation to simplify the calculations and find that the forms are supported on a certain subspace of the full cotangent space. Some results from reference [27] used in our calculations are summarized in appendix A.

A *p*-form ω is closed iff

$$d\omega = 0 \tag{2.1}$$

where d is given locally by $d\theta^{\alpha i}\partial_{\alpha i} + dx^a\partial_a$. For notational convenience, we will replace the basis forms $d\theta^{\alpha i}$ and dx^a by new variables $s^{\alpha i}$ and ψ^a . To automatically incorporate the super-anti-commutative nature of the wedge product, we take the $\{s^{\alpha i}\}$ to be bosonic spinor variables and the $\{\psi^a\}$ to be fermionic vector variables

$$s^{\alpha i}s^{\beta j} = s^{\beta j}s^{\alpha i}, \quad s^{\alpha i}\psi^a = \psi^a s^{\beta j}, \quad \psi^a\psi^b = -\psi^b\psi^a.$$

$$(2.2)$$

The super-*p*-form $\omega = \sum_{s=0}^{p} \omega_{(s,p-s)}$ splits up as a sum of superfield components of degree $s | (p-s)^3$

$$\omega_{(s,p-s)} := \omega_{s\dots s\psi\dots\psi} := s^{\alpha_1 i_1} \dots s^{\alpha_s i_s} \psi^{a_1} \dots \psi^{a_{p-s}} \omega_{\alpha_1 i_1\dots\alpha_s i_s a_1\dots a_{p-s}}, \tag{2.3}$$

³We are borrowing a compact, index-free notation from the theory of ordinary tensor fields on manifolds in which the subscript indicates the vector field on which the corresponding index is contracted (e.g. ∇_X for the directional derivative along the vector field X).

where the *p*-form components $\omega_{\alpha_1 i_1 \dots \alpha_s i_s a_1 \dots a_{p-s}} = \omega_{\alpha_1 i_1 \dots \alpha_s i_s a_1 \dots a_{p-s}}(x, \theta)$ are ordinary N = (1, 0) superfields. The differential *d* splits into two differentials ∂_s and ∂_{ψ}

$$d = \partial_s + \partial_\psi := s^{\alpha i} \partial_{\alpha i} + \psi^a \partial_a := s^{\alpha i} \frac{\partial}{\partial \theta^{\alpha i}} + \psi^a \frac{\partial}{\partial x^a}, \qquad (2.4)$$

and the closure condition fans out into a collection of conditions respecting the grading by number of $s^{\alpha i}$ -type variables. Thus, the closure condition may be represented compactly as

$$s!(d\omega)_{ss\dots\psi\psi\dots} = s\partial_s\omega_{s\dots\psi\psi\dots} + (-1)^s(p+1-s)\partial_\psi\omega_{ss\dots\psi\dots} = 0$$
(2.5)

with the s in front of the first term denoting the total number of s in that term and with p denoting the total degree of the p-form ω .⁴

The differentials ∂_s and ∂_{ψ} do not commute with supersymmetry transformations. To get supersymmetrically covariant *p*-form components, we must replace coordinate derivatives by supercovariant derivatives. In terms of flat supercovariant derivatives D_s and $D_{\psi} = \partial_{\psi}$, the closure condition acquires a flat-space torsion term $T_{ss}^a = 2is^i \gamma^a s_i = 2is^{\alpha i} (\gamma^a)_{\alpha\beta} s_i^{\beta}$:

$$sD_s\omega_{s\dots s\psi\dots\psi} + (-1)^s(p+1-s)\partial_{\psi}\omega_{s\dots s\psi\dots\psi} - i(-1)^s s(s-1)\omega_{s\dots s\gamma(s,s)\psi\dots\psi} = 0, \quad (2.6)$$

where the notation $\omega_{\gamma(s,s)...}$ is shorthand for the contraction of the null vector $\gamma^a(s,s) := s^i \gamma^a s_i$ on the component $\omega_{a...}$ (cf. (A.14)). In the curved superspace version of this closure relation, there are additional torsion terms

$$s\mathcal{D}_{s}\omega_{s...s\psi...\psi} + (-1)^{s}(p+1-s)\mathcal{D}_{\psi}\omega_{s...s\psi...\psi} - i(-1)^{s}s(s-1)\omega_{s...s\gamma(s,s)\psi...\psi} + (-1)^{s}s(p+1-s)T_{s\psi}{}^{\alpha i}\omega_{\alpha is...s\psi...\psi} - \frac{1}{2}(p+1-s)(p-s)T_{\psi\psi}{}^{\alpha i}\omega_{\alpha is...s\psi...\psi} = 0.$$
(2.7)

These torsions, and the six-dimensional curved supergeometry in general, are reviewed in appendix A. In section 4 we will be solving this equation for each $p = 1, \ldots, 6$.

Superforms with no restrictions on their components yield *reducible* representations of the super-Poincaré algebra. Irreducible representations result by setting to zero one or more irreducible components. Once this is done, the closure conditions become non-trivial consistency conditions on the other components (e.g. all components of lower dimension are required to vanish). Following common superspace practice, we will refer to the resulting closure conditions as "Bianchi identities". Thus, to begin solving the Bianchi identities of any particular p-form, we must locate the component of lowest engineering dimension that does not vanish.

Let n denote the largest value of s such that the s|(p-s)-component of the p-form ω is non-vanishing. Then equation (2.7) simplifies to

$$(n+1)\mathcal{D}_s\omega_{\underbrace{s\dots s}_n}\underbrace{\psi\dots\psi}_{p-n} = i(-1)^{s+1}n(n+1)\omega_{s\dots s\gamma(s,s)\psi\dots\psi}.$$
(2.8)

⁴Whenever we use the letter s for a number, we mean the number of $s^{\alpha i}$ -type variables appearing in the relevant formula. For example, the component $s!(d\omega)_{\alpha i\beta j\gamma kde}$ for a 4-form ω has s = 3 and p = 4 giving $3\partial_s \omega_{ss\psi\psi} - 2\partial_{\psi}\omega_{sss\psi}$. We will simplify such formulæ by multiplying by symmetry factors to cancel denominators, as we have done here.

If we further take the projection $s^{\alpha i} \mapsto \lambda^{\alpha} v^i$ to the product of an unconstrained bosonic spinor λ of SU(4) and an unconstrained bosonic spinor v of SU(2), then the right-hand side vanishes and we find that

$$Q\omega_{\underbrace{\lambda\dots\lambda}_{n}\underbrace{\psi\dots\psi}_{p-n}} = 0.$$
(2.9)

Here,

$$Q := \mathcal{D}_{\lambda \otimes v} = \lambda^{\alpha} v^{i} \mathcal{D}_{\alpha i} =: \lambda^{\alpha} \mathcal{D}_{\alpha}^{+}$$
(2.10)

stands for the projected superspace derivative.

The operator Q appears repeatedly in the analysis that follows. Its importance derives from the fact that the condition (2.9) holds for the lowest non-vanishing component of any p-form and, as such, it appears as a defining condition on the superfield from which all other components are derived.⁵ We call the projection $s \mapsto \lambda \otimes v$ the *pure spinor projection* for reasons discussed in appendix A (cf. eq. (A.16) and the surrounding discussion).

Superfields $\Phi = \Phi(x, \theta, v)$ satisfying $\mathcal{D}^+_{\alpha} \Phi = 0$ are known in the projective superspace literature as *analytic* superfields.⁶ When Φ is homogeneous of degree n in the variables v, Φ is said to have *homogeneity weight* n. When we wish to indicate this explicitly, we will do so with a superscript Φ^{+n} . In these terms, equation (2.9) says that the pure spinor projection $\omega^{+n}_{\lambda\ldots\lambda\psi\ldots\psi}$ of the lowest-dimension, non-vanishing component of the p-form is an analytic superfield with homogeneity weight n. The dimension of this field is $d = \frac{1}{2}n + p - n = p - \frac{n}{2}$.

It is possible for the aforementioned projection of the n|(p-n)-component to vanish.⁷ When this happens, the pure spinor projection of the lowest dimension $(d + \frac{1}{2})$ Bianchi identity is trivially satisfied. Passing to the next Bianchi identity (that with dimension d+1) we find that it is the (n-1)|(p-n+1)-component of ω that projects to an analytic superfield provided this projection does not also vanish identically. If it does, we pass to the next Bianchi identity. We proceed this way until we find a value $n' \leq n$ such that the projection of the n'|(p-n')-component does not vanish under pure spinor projection and, therefore, defines an analytic superfield with homogeneity weight n'.⁸ In our analysis, we will find that this field is a *superconformal primary field*, that is, a superfield transforming homogeneously under super-Weyl transformations as we recall in section 3.1 (cf. eq. (3.9)).

In section 3.1, we will also show that $Q^2 = 0$ on superfields defined over the pure spinor subspace. These superfields, graded by homogeneity weight and spin, form a complex of spaces with differential Q. Assuming this, we conclude that differential forms restricted to the pure spinor subspace are sourced by superconformal primary superfields in the cohomology of this complex. In the next section, we will study large families of such superfields.

With the lowest non-vanishing component in hand, the remaining components of the superform can be reconstructed by the usual method. To wit, one first solves the lowest non-trivial Bianchi identity for the dimension- $(d + \frac{1}{2})$ component by inverting the constant tor-

⁵This helps to explain the observation in reference [41] that for differential forms their "spinor cohomology" of coincides with the pure spinor cohomology.

 $^{^{6}\}mathrm{We}$ are glossing over some subtleties here that we address as needed.

⁷This happens, for example, for the p = 3 form H where n = 2 (i.e. $H_{ss\psi} \neq 0$) but $H_{\lambda\lambda\psi} \equiv 0$ (cf. section 4.3).

⁸In the p = 3 example, n = 2 but n' = 1 (cf. section 4.3).

sion $2i(s^i\gamma^a s_i)$ in equation (2.8). These components suffice to solve the next Bianchi identity (2.7) for the dimension-(d+1) component, provided certain additional constraints are imposed on the defining field. This process continues to define the next-higher component and, in principle, additional constraints, until we have reached the top component of dimension p. The final two Bianchi identities must now be identically satisfied [8, 37]. In section 4, we will carry out this procedure to find the components of all the differential forms in curved superspace and verify that the final two identities are satisfied identically in the flat limit.

3 Conformal symmetry

Consider a superfield $\Phi^{\beta_1...\beta_c}{}_{\alpha_1...\alpha_s k_1...f_f}(x,\theta)$ with c symmetric fundamental spinor indices, s symmetric anti-fundamental spinor indices, and f symmetric isospin indices. We introduce the commuting variables $\{\lambda^{\alpha}, \bar{\lambda}_{\alpha}, v^i\}$ in the $\{(4, 1), (\bar{4}, 1), (1, 2)\}$ representations of $SU(4) \times SU(2)$ and replace the superfield with

$$\Phi_{\underbrace{\lambda\dots\lambda\bar{\lambda}\dots\bar{\lambda}}_{s}c}^{+f} = v^{i_1}\dots v^{i_f}\lambda^{\alpha_1}\dots\lambda^{\alpha_s}\bar{\lambda}_{\beta_1}\dots\bar{\lambda}_{\beta_c}\Phi^{\beta_1\dots\beta_c}{}_{\alpha_1\dots\alpha_s i_1\dots i_f}.$$
(3.1)

Lorentz-irreducibility requires tracelessness on pairs of fundamental and anti-fundamental spinor indices. We impose this by requiring

$$\lambda^{\alpha} \bar{\lambda}_{\alpha} = 0. \tag{3.2}$$

Introducing the conjugate momenta $\{\omega_{\alpha}, \bar{\omega}^{\alpha}, p_i\}$ allows us to rewrite the action of the Lorentz and isospin generators as

$$\mathbf{m}_{ab} = -\frac{1}{2}\lambda\gamma_{ab}\omega - \frac{1}{2}\bar{\lambda}\tilde{\gamma}_{ab}\bar{\omega} \quad \text{and} \quad \mathbf{j}_{ij} = v_{(i}p_{j)}.$$
(3.3)

This representation preserves the constraint (3.2). The other derivations preserving the constraint are

$$\mathbf{p}_a = \bar{\lambda} \tilde{\gamma}^a \omega, \quad \mathbf{\mathfrak{t}}^a = \lambda \gamma^a \bar{\omega}, \quad \Delta = \frac{1}{2} \lambda^\alpha \omega_\alpha - \frac{1}{2} \bar{\lambda}_\alpha \bar{\omega}^\alpha, \quad \text{and} \quad \mathbf{q} = v^i p_i. \tag{3.4}$$

Together, they generate the conformal algebra $\mathfrak{so}_{6,2}$

$$\begin{split} [\mathfrak{m}_{ab},\mathfrak{m}_{cd}] &= -2\eta_{c[a}\mathfrak{m}_{b]d} + 2\eta_{d[a}\mathfrak{m}_{b]a}, \qquad \qquad [\mathfrak{p}_{a},\mathfrak{k}^{b}] = 2\delta_{a}^{b}\Delta - 2\mathfrak{m}_{a}^{b}, \\ [\mathfrak{m}_{ab},\mathfrak{p}_{c}] &= -2\eta_{c[a}\mathfrak{p}_{b]}, \qquad \qquad [\Delta,\mathfrak{p}_{a}] = -\mathfrak{p}_{a}, \\ [\mathfrak{m}_{ab},\mathfrak{k}_{c}] &= -2\eta_{c[a}\mathfrak{k}_{b]}, \qquad \qquad [\Delta,\mathfrak{k}_{a}] = \mathfrak{k}_{a}, \end{split}$$
(3.5)

and a decoupled \mathfrak{u}_2 .

An irreducible (iso-)spin-tensor (3.1) is an irreducible representation of this algebra. The derivation Q acts on such representations. In general, its square is proportional to Lorentz and isospin generators (cf. eq. (A.18)). When acting on the representations (3.1), however, the Lorentz $M \to \mathfrak{m}$ and isospin generators $J \to \mathbf{j}$ are represented by (3.3). Simple Fierz rearrangement then implies that the only remaining term is proportional to the constraint (3.2). Therefore,

$$Q^2 = 0 \tag{3.6}$$

when acting on the family of fields of the form (3.1). Thus, these fields form a differential complex graded by spin and isospin.

3.1 Superconformal primary superfields

In this subsection, we use the pure spinor/ambi-twistor-like representation just introduced to construct large families of superconformally covariant field representations by imposing super-Weyl-invariant constraints. The ambi-twistor-like variables are not necessary to define the constraints but they greatly simplify the proof of their super-Weyl covariance. When the resulting representations are on shell, their dynamics are superconformally invariant.

In appendix A we recall the action of super-Weyl transformations (cf. eq. (A.9)–(A.12) preserving the algebra of covariant derivatives defining the six-dimensional, N = (1,0) supergravity theory studied in reference [27]. These transformations are parameterized by a real, unconstrained scalar superfield $\sigma(z)$. In keeping with superspace terminology, we define an irreducible superfield (3.1) to be a Weyl tensor of weight w provided that under such a transformation

$$\delta \Phi = w \sigma \Phi. \tag{3.7}$$

On such fields, the differential transforms as

$$\delta Q = \frac{1}{2}\sigma Q + \frac{1}{2} \left(3s + c - 4f \right) (Q\sigma), \tag{3.8}$$

as follows from (A.9) evaluated on the representation (3.3). Therefore, if Φ is a Weyl tensor with weight w then $Q\Phi$ will be a Weyl tensor of weight $w + \frac{1}{2}$ provided⁹

$$w = 2f - \frac{3}{2}s - \frac{1}{2}c. \tag{3.9}$$

In particular, it is consistent to constrain

$$Q\Phi = 0$$
 provided $w = 2f - \frac{3}{2}s - \frac{1}{2}c.$ (3.10)

Important examples are given in the following table:

1	0 4	þ	(f, s, c)	superfield	${\rm field\ strength/potential}$	w	section
-	1 <i>I</i>	4	(1, 1, 0)	$A_{\alpha i}$	potential	1/2	4.1
4	2 И	V	(1, 0, 1)	$W^{\alpha i}$	field strength	3/2	4.2
4	$2 \mid V$	7	(1, 0, 1)	$V^{\alpha i}$	potential	3/2	4.3.1
	3 0	2	(2, 1, 1)	C_{abij}	potential	2	4.4.1
į	5 F	K	(2, 0, 0)	K_{ij}	field strength	4	4.5
(3 1	5	(3, 1, 0)	$L_{\alpha ijk}$	field strength	9/2	4.6

As we will discuss in detail in section 4, they represent, respectively, the gauge 1-form potential $A := \lambda^{\alpha} v^i A_{\alpha i}$, its 2-form field strength $W := \bar{\lambda}_{\alpha} v_i W^{\alpha i}$, the gauge 2-form potential $V := \bar{\lambda}_{\alpha} v_i V^{\alpha i}$, the gauge 3-form potential $C := (\lambda \gamma^{ab} \bar{\lambda}) v^i v^j C_{ab ij}$, the so-called *linear*

⁹The parameter $\mathcal{D}_{\alpha i}\sigma$ is the S-supersymmetry parameter. Canceling this term implies that the field Φ is invariant under S-supersymmetry, that is, it is a superconformal primary field.

multiplet $K := v^i v^j K_{ij}$ related to the projective Lagrangian density,¹⁰ and the 6-form field strength $L := \lambda^{\alpha} v^i v^j v^k L_{\alpha i j k}$.

In addition to this family of representations, there is an infinite family of symmetric, traceless "spin- ℓ " superfields $J_{c_1...c_\ell i_1...i_f} = J_{(c_1...c_\ell)(i_1...i_f)} - \text{traces: using } \mathfrak{k}$ (cf. eq. (3.4)), let

$$J_f^{(\ell)} := v^{i_1} \dots v^{i_f} \mathfrak{k}^{c_1} \dots \mathfrak{k}^{c_\ell} J_{c_1 \dots c_\ell i_1 \dots i_f}.$$
(3.11)

Then the condition

$$QJ_f^{(\ell)} = 0 \quad \Rightarrow \quad w = 2f - \ell, \tag{3.12}$$

defines a Weyl superfield of weight $2f - \ell$. We will see in section 4.4 that for $\ell = 1$ and f = 2, this condition defines the weight w = 3 field $G_{a\,ij}$ sourcing the 4-form field strength.

Finally, there are seven other families of Weyl superfields that are described naturally in various alternative "polarizations" of the pure spinor variables. To illustrate what we mean by this, we will explicitly work out the only example used in our analysis of differential forms: consider a superfield with s = 0 but c and f arbitrary. Performing the canonical transformation from $\{\bar{\lambda}, v\}$ variables to $\{\omega, p\}$ on the superfield $\Phi \mapsto \Phi^{\dagger}$ only, results in a re-"normal"-ordering under which the conformal weight changes as $w = 2v^i p_i - \frac{1}{2}\bar{\lambda}_{\alpha}\bar{\omega}^{\alpha} - \frac{3}{2}\lambda^{\alpha}\omega_{\alpha} \mapsto 2[-f-2] - \frac{1}{2}[0] - \frac{3}{2}[-c-4]$. In the new polarization, the constraint $Q\Phi^{\dagger} = 0$ is equivalent to the condition¹¹

$$\mathcal{D}_{\alpha i} \Phi^{\alpha \beta_1 \dots \beta_{c-1} \, i j_1 \dots j_{f-1}} = 0 \quad \Rightarrow \quad w = -2f + \frac{3}{2}c + 2 \tag{3.13}$$

on the contraction of the operator $\mathcal{D}_{\alpha i}$ with the spin and isospin indices on $\Phi^{\alpha i...}$. This constraint is compatible with the condition (3.10) when the weights agree, that is, for c = 2f - 1 and $w = f + \frac{1}{2}$. As we will derive in section 4.2, the lowest-weight member of this tower is the the gauge 1-form field strength superfield W. Note that we are not required to impose this condition on such a weight- $(f + \frac{1}{2})$ field. If we do not, we find that the component $Q\Phi^{\dagger}$ is another Weyl tensor of weight f+1. We will see in section 4.3.1 that this observation provides the link between the description of the 2-form gauge potential described by the supefield V and that in terms of its 3-form field strength. The latter is built on the Weyl-weight-2, real scalar superfield Φ related to V by $\Phi = \mathcal{D}_{\alpha i} V^{\alpha i}$ (cf. eq. (4.34)).

4 The super-differential complex

In this section, we will explicitly go through the steps outlined at the end of section 2 for solving the closure conditions (referred to as Bianchi identities) for *p*-forms with $p = 1, \ldots, 6$ subject to the condition that certain components vanish (e.g. $F_{ss} = 0$, $H_{sss} = 0$). In the process, we will find that for each $p \leq 5$, there is an additional constraint on the defining superfield necessary for the closure of the dimension-*p* Bianchi identity. To pass to the

¹⁰The projective Lagrangian density [27] is a homogeneity weight-2, analytic superfield $\hat{K}^{++}(\zeta)$ containing an infinite number of ordinary superfields. Of these, K_{ij} are the first three terms in an expansion in $\zeta := v^2/v^1$ with all other superfields containing only auxiliary components.

¹¹Equivalently, we can keep Φ and perform the transformation on $Q \mapsto Q^{\dagger}$. Then Q^{\dagger} will act by contraction of the form Φ on the vector $\mathcal{D}_{\lambda \otimes v}$.

next p-form in the complex, we relax this last condition, thereby obstructing the closure of the p-form field strength. In doing so, we find the defining superfield for the (p + 1)-form. In this sense, the entire complex is derived from the constraint $F_{\alpha i\beta j} = 0$ imposed on the 2-form. The lowest non-vanishing components of the resulting forms have precisely the dimensions found to imply superconformal invariance in section 3.1.

The result for flat forms with p = 2, ..., 5 is represented schematically in the following table:



In order to fit the entries into the table, we have suppressed the 1- and 6-form, are ignoring numerical factors, and we use * in the higher components of the forms of degree p = 4, 5 to schematically denote the Hodge dual.

Very roughly, going up a *p*-form chain corresponds to applying the operator $D_i \tilde{\gamma}_{\psi} \partial/\partial s_i$. Going across corresponds to finding the dimension-*p* component and replacing the *DD* operator acting on the defining field with some field that is bilinear in *s*. In fact, this is happening because the Bianchi identity has *reducible* Lorentz/isospin components while the *DD* field strength of the *p* form is irreducible. Imposing that the additional irreducible components vanish closes the *p*-form Bianchi identity $d\omega_p = 0$. Alternatively, we can interpret this additional irreducible part as the lowest non-trivial component of a (p+1)-form ω_{p+1} . Then the statement is that the non-vanishing of this new form is the obstruction to the *p*-form Bianchi identity, that is, $d\omega_p = \omega_{p+1}$.

For example, starting with $F_{ss} = 0$ and working our way up to the Bianchi identity $(dF)_{ss\psi} = 0$, we find that $F_{\psi\psi} \propto D^i \tilde{\gamma}_{\psi\psi} W_i$ for the top component of the 2-form. However, in that same identity, there remains uncanceled the Lorentz-irreducible term $D_{\alpha i} W^{\alpha i}$ which the Bianchi identity sets to zero. Alternatively, we may decide to deform the closure

condition by introducing a source $H_{ss\psi}$ at this level. Then the new identity is dF = H. Consistency requires dH = 0, which we then proceed to solve. But this is just the Bianchi identity for the 3-form as it appears in the second column of the table.

Below the dividing line in the table are the conditions the defining superfields satisfy. The top line represents the relation $Q\Phi = 0$ on the pure spinor cone. The line below it denotes additional conditions required for the closure of the Bianchi identity for that particular *p*-form. The interpretation of the (p+1)-form as an obstruction to the *p*-form Bianchi identity is reflected in the fact that the left-hand side of each condition on the bottom line is in the same irreducible Lorentz and isospin representation as the defining field to the right of it.¹² In the following subsections, we will make all of these statements explicit.

We should mention that we are not claiming that the set of *p*-form representations we obtain is complete;¹³ in lower dimensions, it is possible to have so-called "variant representations" [1, 3, 42]. However, the tower we obtain is uniquely determined by working our way up from the constraint $F_{ss} = 0$. Furthermore, each form has a superconformal primary field as its lowest non-vanishing component. Taken together, it may be that the resulting complex is unique. Proving this or finding counter-examples (variant representations) should be possible by first classifying all superconformal constraints of scaling weight $w \leq p$ along the lines of section 3.1 and then inspecting them for proper Lorentz and iso-spin structure.¹⁴

The complex described in this section was recently interpreted as the Chevalley-Eilenberg complex of *p*-co-chains of the supersymmetry algebra with values in the space of superfields [8], with a differential $\omega \mapsto d_{CE}\omega$ that is defined by the left-hand-side of equation (2.7) or, for simplicity, its flat-space analogue (2.6).¹⁵ Let us denote the lowestdimension non-vanishing component of ω by ϕ (e.g. for the 2-form $\omega = F$ of section 4.2, $\phi = W$ enters in eq. (4.10) as the spinor fieldstrength.) The form of ϕ is determined by the Bianchi identities. Similarly, the higher Bianchi identities define the remaining components of ω in terms of superspace derivatives on ϕ projected to combinations of γ -matrices (and in curved space, covariantized with torsions) that may or may not be Lorentz-irreducible. The projection to the complementary representations define constraints on ϕ .

This picture was made precise in reference [8] (see also [37]) in the following sense: when the coefficients in the complex are taken to be constant superfields, we obtain an algebraic complex with a reduced differential d_{CE0} which is essentially the contraction $\psi \mapsto \gamma(s, s)$ (a.k.a. interior product with the vector defined by s and the Clifford map). Then the set of constraints on ϕ is in bijection with the cohomology of this reduced differential d_{CE0}

¹²For example, for p = 2 the spinor field strength has no scalar component $D_i W^i$: this term appears in the Bianchi identity at the same level as $F_{\psi\psi}$ but cannot be canceled by it since there is no Lorentz-invariant way to absorb a scalar into a 2-form. This component must therefore be set to vanish if we want the Bianchi identity for the 2-form to be satisfied. This scalar superfield is of the same form as the defining field Φ in the 3-form to the right of it in the table.

¹³We thank Gabriele Tartaglino-Mazzucchelli for raising this question.

¹⁴For example, H_{sss} has scaling weight $\frac{3}{2}$. There is a representation of this dimension which we call $V^{\alpha i}$ but it cannot be used to construct this particular component H_{sss} since, for example, $s^{\alpha i}s_i^{\beta}s^{\gamma j}\varepsilon_{\alpha\beta\gamma\delta}V_j^{\delta}\equiv 0$. Instead, it enters into $H_{ss\psi}$ as $\mathcal{D}_{\alpha i}V^{\alpha i}$, as explained in section 4.3.1.

 $^{^{15}}$ The analysis of [8, 37] was carried out for the flat five-dimensional case but the results generalize to other dimensions.

and the set of equations defining the components of ω in terms of ϕ is bijective to the co-boundaries of the reduced Chevalley-Eilenberg complex. An immediate corollary of this theorem is the well-known statement that there are no new constraints imposed by the last two Bianchi identities because the reduced cohomology is empty there (since co-boundaries must have at least two *s* variables). We will generally not use the results just described (except to avoid a particularly tedious and unilluminating calculation in section 4.4) as the analysis presented here predates that of [8, 37].

4.1 The closed 1-form

The components of a generic 1-form are (A_s, A_{ψ}) . The first condition for a closed 1-form is the component of (2.7) with the most ss:

$$2(dA)_{ss} = 2\mathcal{D}_s A_s - 2iA_{\gamma(s,s)} = 0.$$
(4.1)

The pure spinor projection of this equation is simply (2.9)

$$QA = 0 \tag{4.2}$$

where $A = \lambda^{\alpha} v^i A_{\alpha i}$ defines a Weyl tensor superfield provided $w = \frac{1}{2}$ (3.10). This condition, which is equivalent to $\mathcal{D}_{(\alpha(i}A_{\beta)j)} = 0$, was solved in flat space in references [43] and [44] based on the four-dimensional, N = 2 solution of Mezinçescu [45] as $A_{\alpha i} = D_{\alpha i}U + D_{\alpha}^{j}U_{ij}$. Substituting this back into the pure spinor constraint QA = 0, we find that the isotriplet prepotential U_{ij} is required to satisfy $D^k \tilde{\gamma}_{abc} D_k U_{ij} = 0$. Due to the flat-space identity $D^m \tilde{\gamma}_{abc} D_m D_{ijkl}^4 \equiv 0$, this implies that it can be written in terms of the unconstrained Mezinçescu prepotential u^{ij} as $U_{ij} = D_{ijkl}^4 u^{kl}$.

Returning to curved space, the vector component the super-1-form is determined by the dimension-1 Bianchi identity to be

$$A_{\psi} = -\frac{i}{8} \mathcal{D}^k \tilde{\gamma}_{\psi} A_k. \tag{4.3}$$

Substituting the curved-space analogue of the solution above

$$A_{\alpha i} = \mathcal{D}_{\alpha i} U + \mathcal{D}_{\alpha}^{j} U_{ij} \quad \text{with} \quad \mathcal{D}^{k} \tilde{\gamma}_{abc} \mathcal{D}_{k} U_{ij} + 256 i N_{abc}^{(-)} U_{ij} = 0, \tag{4.4}$$

this becomes $A_{\psi} = \mathcal{D}_{\psi}U - \frac{i}{8}\mathcal{D}_{(i}\tilde{\gamma}_{\psi}\mathcal{D}_{j)}U^{ij}$. The U part drops out of the dimension- $\frac{3}{2}$ Bianchi identity

$$\mathcal{D}_s A_\psi - \mathcal{D}_\psi A_s - T_{s\psi}{}^{\alpha i} A_{\alpha i} = 0 \tag{4.5}$$

which constrains the prepotential U_{ij} by $s_i \gamma_{\psi} W_U^i = 0$ for

$$W_U^{\alpha i} := \frac{1}{8} \mathcal{D}^{3\alpha i j k} U_{jk} - \frac{i}{6} (\tilde{\gamma}^a)^{\alpha \beta} \mathcal{D}_a \mathcal{D}_{\beta j} U^{ij} + \text{ torsion terms.}$$
(4.6)

Note that this is the form of the field strength superfield for a gauge 1-form with prepotential U_{ij} . In the next section, we will relax the condition forcing it to vanish, thereby generating the closed 2-form field strength as an obstruction to the closure of the 1-form being worked out here. Setting the field strength superfield $W_U = 0$ is gauge equivalent to setting $U_{ij} = 0$. Doing this, we find that the closed 1-form is given by $(A_s, A_{\psi}) = (\mathcal{D}_s U, \mathcal{D}_{\psi} U)$. The dimension-2 Bianchi identity

$$2\mathcal{D}_{\psi}A_{\psi} - \frac{1}{2} \cdot 2T_{\psi\psi}{}^{\alpha i}A_{\alpha i} \equiv 0 \tag{4.7}$$

is satisfied identically. We conclude that, as expected, the unique closed, Weyl-covariant 1-form is the exact 1-form with weight-0 (iso-)scalar potential U.

4.2 The closed 2-form

The closed 1-form A of the previous section satisfied two constraints. The first of these was the pure spinor constraint QA = 0 (4.2). The second was the dimension- $\frac{3}{2}$ Bianchi identity (4.5) constraining the prepotential U to vanish. We can deform this particular Bianchi identity by introducing the superfield 2-form field strength $F_{s\psi} = (dA)_{s\psi}$ as an obstruction to the closure of the 1-form field A. In this interpretation, continuing to impose the pure spinor condition (4.2) corresponds to keeping $F_{ss} = 0$. In fact, the two conditions are equivalent: generally, $F_{\alpha i\beta j} = F_{\beta j\alpha i} = F_{(\alpha (i\beta)j)} + F_{[\alpha [i\beta]j]}$ but the second term is equivalent to $\varepsilon_{ij}(\gamma^a)_{\alpha\beta}F_a$ which can be absorbed by a field redefinition into the vector component A_a . The remaining term $F_{(\alpha (i\beta)j)} = \mathcal{D}_{(\alpha (i}A_{\beta)j)}$ is precisely the combination QA.

In this section, we will solve the Bianchi identities for the closed 2-form F subject to

$$F_{ss} = 0. \tag{4.8}$$

We emphasize that this is the only input from section 4.1 that we will use. The lowest-level Bianchi identity relates the components F_{ss} and $F_{s\psi}$ as

$$3\mathcal{D}_s F_{ss} + 3 \cdot 2iF_{s\gamma(s,s)} = 0. \tag{4.9}$$

By the Fierz identity (A.15), the condition $F_{ss} = 0 \Rightarrow F_{s\gamma(s,s)} = 0$ is compatible with the solution

$$F_{s\psi} = 2i(s^i \gamma_{\psi} W_i), \tag{4.10}$$

for some positive chirality, fermionic superfield $W^{\alpha i}$. The dimension-2 identity reads

$$2\mathcal{D}_s F_{s\psi} + \mathcal{D}_{\psi} F_{ss} - 2i F_{\gamma(s,s)\psi} = 0. \tag{4.11}$$

The pure spinor projection of this implies

$$QW = 0 \tag{4.12}$$

in the sense of section 3.1. To see this, note that under pure spinor projection, $(s^i \gamma_{\psi} W_i) \mapsto c_{\alpha} W^{\alpha+}$ where we define $c_{\alpha} := (\lambda \gamma_{\psi})_{\alpha}$. Although this contravariant spinor is fermionic, it satisfies $\lambda^{\alpha} c_{\alpha} \equiv 0$ analogously to (3.2) so that the analysis of that section remains valid in this special parameterization. The pure spinor condition (4.12) defines a Weyl tensor superfield provided its weight is $w = \frac{3}{2}$. By condition (3.2) or, rather, its fermionic version, it is equivalent to

$$\mathcal{D}_{(i}\tilde{\gamma}_{ab}W_{j)} = 0. \tag{4.13}$$

As a practical matter, this condition says that at the level of component field strengths, there is no triplet of 2-forms.

The pure spinor projection isolates the isospin triplet part of the dimension-2 Bianchi identity (4.11). The remaining isospin singlet part contains a superfield 2-form term and a scalar term. Canceling the 2-form part results in the definition of the top component of the closed 2-form

$$F_{\psi\psi} = -\frac{1}{4} \mathcal{D}_k \tilde{\gamma}_{\psi\psi} W^k, \qquad (4.14)$$

leaving only $(s^i \gamma_{\psi} s_i)(\mathcal{D}_{\beta j} W^{\beta j}) = 0$ uncanceled. Therefore, in addition to the constraint (4.13), we are required to impose the vanishing of the scalar term

$$\mathcal{D}_{\alpha i} W^{\alpha i} = 0. \tag{4.15}$$

Note that this condition is Weyl invariant by (3.13) since $w = \frac{3}{2}$. Furthermore, it is required for consistency of that part of the dimension-2 Bianchi identity that drops out of the pure spinor projection. By contrast, the constraint (4.12) is defined by the projection to the pure spinor subspace. In the next subsection (cf. section 4.3), we will interpret the lowest 3-form component as an obstruction to this additional condition (4.15).

The constraints (4.13) and (4.15) define the abelian 2-form field strength representation W [46]. Its derivative can be expanded as

$$\mathcal{D}_{\alpha i}W^{\beta j} = \delta_i^j F_\alpha{}^\beta + \delta_\alpha{}^\beta X_i{}^j \tag{4.16}$$

for some iso-triplet superfield $X_{ij} = X_{(ij)}$ whose lowest components are auxiliary fields. The dimension- $\frac{5}{2}$ Bianchi identity

$$\mathcal{D}_s F_{\psi\psi} - 2\mathcal{D}_{\psi} F_{s\psi} - 2T_{s\psi}{}^{\alpha i} F_{\alpha i\psi} = 0 \tag{4.17}$$

serves to relate the field equations to X and imply no additional conditions on W. Likewise, the dimension-3 identity

$$\mathcal{D}_{\psi}F_{\psi\psi} - T_{\psi\psi}^{\alpha\imath}F_{\alpha\imath\psi} = 0 \tag{4.18}$$

implies no new constraints. Consequently, this (5+3)|(4+4)-component representation is off-shell.

4.3 The closed 3-form

The defining superfield W of the abelian 2-form satisfies, in addition to the pure spinor condition (4.12), the dimension-2 constraint (4.15) as follows from the $0 = (dF)_{ss\psi} \propto$ $(s^i \gamma_{\psi} s_i) (\mathcal{D}_{\beta j} W^{\beta j})$ Bianchi identity. In keeping with our general philosophy, we source this equation with

$$H_{ss\psi} = 2i(s^i \gamma_{\psi} s_i)\Phi \tag{4.19}$$

for some Weyl tensor superfield Φ of weight w = 2 [43, 44]. This is consistent with the dimension-2 Bianchi identity

$$4\mathcal{D}_s H_{sss} - 4 \cdot 3i H_{ss\gamma(s,s)} = 0. \tag{4.20}$$

provided we also impose $H_{sss} = 0$. Note that $H_{ss\psi}$ vanishes under the pure spinor projection. This is consistent with the projection of the dimension- $\frac{5}{2}$ Bianchi identity

$$3\mathcal{D}_s H_{ss\psi} - \mathcal{D}_{\psi} H_{sss} + 3 \cdot 2i H_{s\gamma(s,s)\psi} = 0, \qquad (4.21)$$

from which we easily obtain

$$H_{s\psi\psi} = -(s^i \gamma_{\psi\psi} \mathcal{D}_i \Phi). \tag{4.22}$$

Therefore, in the notation of section 2, n = 2 and n' = 1. Plugging this into the dimension-3 identity

$$2\mathcal{D}_s H_{s\psi\psi} + 2\mathcal{D}_{\psi} H_{ss\psi} - 2iH_{\gamma(s,s)\psi\psi} + 4T_{s\psi}{}^{\alpha i}H_{\alpha is\psi} = 0$$
(4.23)

and taking the pure spinor projection, we find the condition

$$\frac{1}{8} (\lambda \gamma_{\psi \psi a} \lambda) \left(\mathcal{D}^+ \tilde{\gamma}^a \mathcal{D}^+ + 16i C^{a++} \right) \Phi = 0.$$
(4.24)

Here, we have defined the combination

$$\mathcal{D}_{a\,ij}^2 := \frac{1}{2} \mathcal{D}_{(i} \tilde{\gamma}_a \mathcal{D}_{j)} \tag{4.25}$$

which will show up repeatedly. Thus, the dimension-3 identity implies the curved 3-form field strength constraint [27]

$$\mathcal{D}_{a\,ij}^2 \Phi + 8iC_{aij} \Phi = 0 \tag{4.26}$$

with the remaining combination defining the 3-form field strength component

$$H_{\psi\psi\psi} = \frac{i}{8} \left(\mathcal{D}^k \tilde{\gamma}_{\psi\psi\psi} \mathcal{D}_k + 128iN_{\psi\psi\psi} \right) \Phi.$$
(4.27)

Acting on the constraint (4.26) with \mathcal{D}_s results in the Dirac equation

$$i\mathcal{D}^{\alpha\beta}\mathcal{D}^{i}_{\beta}\Phi - iC^{\alpha\beta}i^{j}\mathcal{D}_{\beta}j\Phi - 2iN^{\alpha\beta}\mathcal{D}^{i}_{\beta}\Phi - 12i\mathcal{C}^{\alpha i}\Phi = 0$$

$$(4.28)$$

so this multiplet is on-shell.¹⁶ The calligraphic torsion components entering here and below are the dimension- $\frac{3}{2}$ components of the supergravity torsions (A.7). They are included here only for completeness and are not critical to the understanding of the 3-form. Similarly, we find the covariantized Klein-Gordon equation that results from contracting with another spinor derivative:

$$0 = \mathcal{D}^{a}\mathcal{D}_{a}\Phi + 8C^{a}_{ij}C^{ij}_{a}\Phi + \frac{i}{6}N^{abc}\mathcal{D}^{k}\tilde{\gamma}_{abc}\mathcal{D}_{k}\Phi - \frac{3i}{2}\mathcal{D}_{\alpha i}\mathcal{C}^{\alpha i}\Phi - 3i\mathcal{C}^{\alpha i}\mathcal{D}_{\alpha i}\Phi + \frac{5i}{2}\mathcal{N}^{\alpha i}\mathcal{D}_{\alpha i}\Phi.$$

$$(4.29)$$

The higher Bianchi identities

$$\mathcal{D}_s H_{\psi\psi\psi} - 3\mathcal{D}_{\psi} H_{s\psi\psi} - 3T_{s\psi}{}^{\alpha i} H_{\alpha i\psi\psi} - 3T_{\psi\psi}{}^{\alpha i} H_{\alpha is\psi} = 0$$

¹⁶An alternative interpretation of this formula in curved space is as a mechanism for defining geometrical objects in the Weyl multiplet in terms of those in the tensor multiplet [47] (see also section 2.3 of [27]).

$$4\mathcal{D}_{\psi}H_{\psi\psi\psi} - \frac{1}{2} \cdot 4 \cdot 3T_{\psi\psi}{}^{\alpha i}H_{\alpha i\psi\psi} = 0 \qquad (4.30)$$

do not imply any new conditions on Φ beyond those following from the constraint (4.26). Instead of presenting calculations resulting in equations implied by (4.28) and (4.29), we will merely verify that the flat limits are identically satisfied. For the dimension- $\frac{7}{2}$ identity:

$$D_{s}H_{\psi\psi\psi} = \frac{1}{2}(s^{i}\gamma^{a}\tilde{\gamma}_{\psi\psi\psi}D_{i})\partial_{a}\Phi = \frac{1}{2}(s^{i}\gamma^{a}\tilde{\gamma}_{\psi\psi\psi}D_{i} - s^{i}\tilde{\gamma}_{\psi\psi\psi}\gamma^{a}D_{i})\partial_{a}\Phi$$
$$= -3\partial_{\psi}(s^{i}\gamma_{\psi\psi}D_{i}\Phi) = 3\partial_{\psi}H_{s\psi\psi}$$
(4.31)

where we have used the Dirac equation on the spinor of Φ . Similarly,

$$\partial_{\psi} H_{\psi\psi\psi} = -\frac{i}{64} \epsilon_{\psi\psi\psi\psi}{}^{ab} (D^{i} \tilde{\gamma}_{abc} D_{i}) \partial^{c} \Phi$$

$$= -\frac{i}{64} \epsilon_{\psi\psi\psi\psi}{}^{ab} D^{i} (\tilde{\gamma}_{ab} \tilde{\gamma}_{c} - 2\eta_{c[a} \tilde{\gamma}_{b]}) D_{i} \partial^{c} \Phi$$

$$= -\frac{i}{32} \epsilon_{\psi\psi\psi\psi}{}^{ab} D^{i} \tilde{\gamma}_{a} D_{i} \partial_{b} \Phi = \frac{1}{4} \epsilon_{\psi\psi\psi\psi}{}^{ab} \partial_{a} \partial_{b} \Phi \equiv 0, \qquad (4.32)$$

by using the Dirac equation in the third equality.

4.3.1 Alternative formulation

The recovery of the 3-form superfield as an obstruction to one of the two defining conditions of the vector muliplet (4.15), suggests an alternative description of this form in terms of a dimension- $\frac{3}{2}$ spinor superfield $V^{\alpha i}$ [48]. As we are obstructing only the scalar constraint, this field would still be required to satisfy the defining condition analogous to (4.12) or (4.13):

$$QV = 0 \quad \Leftrightarrow \quad \mathcal{D}_{(i}\tilde{\gamma}_{ab}V_{j)} = 0.$$
 (4.33)

It can be shown by brute force calculation that this condition is equivalent to the constraint (4.26) with

$$\Phi = \mathcal{D}_{\alpha i} V^{\alpha i}.\tag{4.34}$$

This description neatly incorporates the gauge invariance of the 3-form field strength since Φ is invariant under $\delta V^{\alpha i} = \Lambda^{\alpha i}$ where Λ satisfies both constraints (4.13) and (4.15) defining the vector multiplet. The condition on V implies that

$$\mathcal{D}_{\alpha i} V^{\beta j} = \frac{1}{8} \delta^j_i \delta^\beta_\alpha \Phi + \frac{1}{4} \delta^j_i B_\alpha{}^\beta + \delta^\beta_\alpha Y_i{}^j$$
(4.35)

for a scalar Φ , 2-form potential B, and an auxiliary triplet Y. Under the gauge transformation, the 2-form gauge field transforms into the field strength of a 1-form λ as $\delta B = d\lambda$.

4.4 The closed 4-form

The closure constraint (4.24) on the 3-form can be obstructed with a 4-form, the lowest non-trivial component of which is given by

$$G_{ss\psi\psi} = (s^i \gamma_{\psi\psi a} s^j) G^a_{ij}. \tag{4.36}$$

This is consistent with the dimension- $\frac{7}{2}$ Bianchi identity

$$3\mathcal{D}_s G_{ss\psi\psi} + 3 \cdot 2iG_{s\gamma(s,s)\psi\psi} = 0 \tag{4.37}$$

under the pure spinor projection, provided

$$(\gamma_{abc})_{(\alpha\beta}\mathcal{D}_{\gamma)(k}G^{c}_{ij)} = 0 \quad \Leftrightarrow \quad \Pi^{c\gamma}_{a\alpha}\mathcal{D}_{\gamma(k}G_{c\,ij)} = 0 \tag{4.38}$$

where Π is the projector onto the γ -traceless subspace of the spinor-vector representation (A.8). To see this, note that the pure spinor projection of (4.36) is proportional to $(c\tilde{\gamma}^a c)G_a^{++}$. The condition that this be annihilated by Q then becomes equivalent to the Weyl-invariant constraint (3.12) for $\ell = 1$ ($\Rightarrow w = 3$). The remaining part of the Bianchi identity (4.37) is, then, easily solved for the dimension- $\frac{7}{2}$ component

$$G_{s\psi\psi\psi} = -\frac{i}{12} \epsilon_{\psi\psi\psi}^{abc} (s^i \gamma_{ab} \mathcal{D}^j) G_{cij}.$$
(4.39)

The dimension-4 Bianchi identity is

$$2\mathcal{D}_s G_{s\psi\psi\psi} + 3\mathcal{D}_{\psi} G_{ss\psi\psi} - 2iG_{\gamma(s,s)\psi\psi\psi} + 2 \cdot 3T_{s\psi}{}^{\alpha i}G_{\alpha is\psi\psi} = 0.$$
(4.40)

The pure spinor projection gives three conditions:

$$\mathcal{D}^2_{a\,k(i}G^{a\,k}_{j)} + 24iC_{a\,k(i}G^{a\,k}_{j)} + 8i\mathcal{D}_a G^a_{\,ij} = 0, \qquad (4.41)$$

$$\mathcal{D}^2_{(a\,k(i}G_{b)\,j)}{}^k - 4i\mathcal{D}_{(a}G_{b)\,ij} + \dots - \text{trace} = 0, \qquad (4.42)$$

$$\mathcal{D}^{2}_{[a\,k(i}G_{b]\,j)}{}^{k} - 4i\mathcal{D}_{[a}G_{b]\,ij} - \frac{1}{8}\mathcal{D}^{k}\tilde{\gamma}_{abc}\mathcal{D}_{k}G^{c}_{ij} + \ldots = 0, \qquad (4.43)$$

where the ellipses stand for unilluminating torsion corrections. The second and third condition follow from the constraint (4.38) by contraction with $(\mathcal{D}^k \tilde{\gamma}_b)^{\alpha}$. Indeed, the combination $\mathcal{D}^2_{a\,k(i}G_{b\,j)}{}^k - 4i\mathcal{D}_aG_{b\,ij} = -\frac{3}{4}\mathcal{D}^k\tilde{\gamma}_a\mathcal{D}_{(k}G_{b\,ij)}$. Since Π is γ -traceless, we get no condition upon contraction with η^{ab} . Therefore, the only condition not already implied by the lower Bianchi identities is the condition (4.41) on the trace. It is this condition that we will source to get the 5-form (cf. section 4.5).

The remaining terms in (4.40) determine the dimension-4 component

$$G_{\psi\psi\psi\psi} = -\frac{1}{24} \epsilon_{\psi\psi\psi\psi}{}^{ab} \left(\mathcal{D}_{a\,ij}^2 - 40iC_{a\,ij} \right) G_b{}^{ij}.$$

$$(4.44)$$

In this calculation, there are no irreducible components beyond this 4-form that need to be canceled so we do not generate any additional constraints on G_{aij} at this level.

The remaining identities are the dimension- $\frac{9}{2}$ identity

$$\mathcal{D}_s G_{\psi\psi\psi\psi} - 4\mathcal{D}_{\psi} G_{s\psi\psi\psi} - 4T_{s\psi}{}^{\alpha i} G_{\alpha i\psi\psi\psi} - \frac{1}{2} \cdot 4 \cdot 3T_{\psi\psi}{}^{\alpha i} G_{\alpha is\psi\psi} = 0 \qquad (4.45)$$

and the dimension-5 identity

$$5\mathcal{D}_{\psi}G_{\psi\psi\psi\psi} - \frac{1}{2} \cdot 5 \cdot 4T_{\psi\psi}{}^{\alpha i}G_{\alpha i\psi\psi\psi} = 0.$$
(4.46)

They should be satisfied identically in the flat limit. This calculation, however, turns out to be particularly tedious. Instead of checking it explicitly, we cite the corollary in [8, 37] stating that the this identity is required to hold due to the vanishing cohomology of a certain algebraic complex. In the flat limit, $(*G)_{ab} = \frac{1}{12}D^{2\ ij}_{[a}G_{b]\ ij}$ and thus it must be that $\partial^b(*G)_{ab} \equiv 0$: the closure of the top component implies that the dual 2-form *G is divergenceless up to torsion terms.

4.4.1 Alternative formulation

In section 4.3.1 we explored the alternative "potential" formulation of the gauge 2-form. There, the condition defining the representation was expressed as QV = 0 (4.33) instead of the condition (4.26) in terms of its field strength Φ . Similarly, one expects to be able to obstruct the closure condition in this potential-type formulation by taking

$$QV = C. (4.47)$$

Since $V = \bar{\lambda}_{\alpha} v_i V^{\alpha i}$ is a field with (f, s, c) = (1, 0, 1) and Q is an operator of type (1, 1, 0), C is of type (2, 1, 1), that is, $C := (\lambda \gamma^{ab} \bar{\lambda}) v^i v^j C_{ab \, ij}$.¹⁷ Consistency then implies

$$QC = 0 \quad \Rightarrow \quad w = 2 \tag{4.49}$$

where the weight, again, follows from the general formula (3.10).

Recall that the vector multiplet field strength W obeys two conditions (4.12) and (4.15). Relaxing (4.15) introduces the potential V for the 3-form field strength which still obeys (4.33). We are now relaxing this second condition by introducing the potential C for the 4-form field strength G = dC. The constraints on G imply that $C_{ss\psi} = s^i \gamma_{\psi}{}^{ab} s^j C_{ab\,ij}$ is the lowest non-vanishing component of this potential. The pure spinor projection of this component with $C_{ab\,ij} := \frac{1}{8} D_{(i} \tilde{\gamma}_{ab} V_{j)}$ gives back equation (4.47).

4.5 The closed 5-form

The obstruction to closure of the 4-form is the left-hand side of (4.41). Our procedure, then, implies that the lowest component of the closed 5-form is given in terms of a superfield K_{ij} by

$$K_{ss\psi\psi\psi} = s^i \gamma_{\psi\psi\psi} s^j K_{ij}. \tag{4.50}$$

This is consistent with the dimension- $\frac{9}{2}$ Bianchi identity

$$3\mathcal{D}_s K_{ss\psi\psi\psi} - 3\mathcal{D}_{\psi} K_{sss\psi\psi} + 3 \cdot 2iK_{s\gamma(s,s)\psi\psi\psi} = 0 \tag{4.51}$$

in the pure spinor projection provided

$$QK = 0 \quad \Leftrightarrow \quad \mathcal{D}_{\gamma(k}K_{ij)} = 0. \tag{4.52}$$

$$D_{a\,ij}^2 \Phi = D_{abc}^2 C_{ij}^{bc} + 4i\partial^b C_{ab\,ij}.$$
(4.48)

¹⁷For purposes of illustration we mention that, in terms of the field strength Φ , the flat-space version of equation (4.47) implies

This condition is Weyl invariant when w = 4 (3.9) in agreement with the engineering dimension of $K_{ss\psi\psi\psi}$. This analyticity constraint implies that

$$\mathcal{D}_{a(i}^{2}{}^{k}K_{j)k} + 24iC_{a(i}{}^{k}K_{j)k} + 4i\mathcal{D}_{a}K_{ij} = 0$$
(4.53)

and

$$\mathcal{D}^k \tilde{\gamma}_{abc} \mathcal{D}_k K_{ij} - 128i N_{abc}^{(-)} K_{ij} = 0.$$
(4.54)

The remaining part of the Bianchi identity defines the dimension- $\frac{9}{2}$ component of K to be

$$K_{s\psi\psi\psi\psi} = -\frac{i}{12} \epsilon_{\psi\psi\psi\psi}{}^{ab} (s^i \gamma_{ab} \mathcal{D}^j) K_{ij}.$$
(4.55)

The dimension-5 Bianchi identity

$$2\mathcal{D}_s K_{s\psi\psi\psi\psi} + 4\mathcal{D}_{\psi} K_{ss\psi\psi\psi} - 2iK_{\gamma(s,s)\psi\psi\psi\psi} - 8(T_{s\psi}{}^i\gamma_{\psi\psi\psi}s^j)K_{ij} = 0$$
(4.56)

is identically satisfied in the pure spinor limit due to the constraint (4.54) and cancelation of the \mathcal{D}_{ψ} terms. The remaining part determines the top component of K to be

$$K_{\psi\psi\psi\psi\psi\psi} = \frac{1}{24} \epsilon_{\psi\psi\psi\psi\psi}^{a} \left(\mathcal{D}_{a\,ij}^{2} - 48iC_{aij} \right) K^{ij}.$$
(4.57)

The dimension- $\frac{11}{2}$ identity is

$$\mathcal{D}_{s}K_{\psi\psi\psi\psi\psi\psi} - 5\mathcal{D}_{\psi}K_{s\psi\psi\psi\psi} - 5T_{s\psi}{}^{\alpha i}K_{\alpha i\psi\psi\psi\psi} - \frac{1}{2}\cdot 5\cdot 4T_{\psi\psi}{}^{\alpha i}K_{\alpha is\psi\psi\psi} = 0.$$
(4.58)

It serves only to define the θ^3 -terms in K in terms of space-time derivatives acting on the lower components and is otherwise unilluminating.

The dimension-6 identity is

$$6\mathcal{D}_{\psi}K_{\psi\psi\psi\psi\psi\psi} - \frac{1}{2} \cdot 6 \cdot 5T_{\psi\psi}{}^{\alpha i}K_{\alpha i\psi\psi\psi\psi} = 0.$$
(4.59)

It tells us that, in the flat limit, the bosonic projection of the top component of the 5-form K is closed in the bosonic sense. We may check this explicitly by using the flat covariant derivative identity $\partial^a D^2_{a\,ij} = \frac{i}{12} D^{3\alpha}_{ijk} D^k_{\alpha}$. It implies that the dual form $*K_a$ is divergenceless

$$\partial^a (*K)_a = 0 \tag{4.60}$$

due to the analyticity constraint (4.52) on K.

Another way to understand this result is by comparison with the 1-form of section 4.1. In the flat limit, the constraint (4.54) agrees with the defining condition (4.4) of a (gauge) 1-form prepotential. This implies that there is a 1-form at the θ^2 -level of K. Since K is a field strength, and due to the dimension of this component, this vector must be a field strength. That this component is divergenceless where that of the vector multiplet was not is a consequence of the stronger constraint (4.52) (from which (4.54) follows).

4.6 The closed 6-form

The top component of the 5-form K defined in (4.57) solves the dimension-5 Bianchi identity with no additional requirements on the superfield K_{ij} beyond the defining pure spinor condition (4.52). As there is no obstruction to the closure of the 5-form, our procedure does not generate a non-vanishing 6-form at this level. (The Chevalley-Eilenberg complex with constant coefficients is exact at this level.)

One possibility is to proceed with the proposal by Drummond and Howe [49] which corresponds to sourcing equation (4.53). (This also suggests the alternative co-dimension 0-form with based on sourcing (4.54).) An action constructed from such an obstruction is of "superaction" type [50]. Alternatively, we may force the violation of the 5-form Bianchi identity by obstructing the defining relation and attempting to interpret the result as a closed 6-form. (This alternative is also suggested by the composite 6-form of section 5.2.3.) This corresponds to the Ansatz

$$L_{sss\psi\psi\psi} = (s^i \gamma_{\psi\psi\psi} s^j) s^{\alpha k} L_{\alpha ijk}$$

$$\tag{4.61}$$

in terms of the weight $w = \frac{9}{2}$ superfield $L_{\alpha ijk}$.¹⁸ Upon pure spinor projection, this gives¹⁹

$$L_{sss\psi\psi\psi} \mapsto \lambda^{\alpha} (\lambda \gamma_{\psi\psi\psi} \lambda) v^{i} v^{j} v^{k} L_{\alpha i j k}.$$

$$(4.62)$$

The projection of the lowest-dimension Bianchi identity (dimension 5)

$$4\mathcal{D}_s L_{sss\psi\psi\psi} - 4 \cdot 3i L_{ss\gamma(s,s)\psi\psi\psi} = 0, \qquad (4.63)$$

as usual, requires QL = 0 which is again a condition of type (3.10). Explicitly,

$$QL = 0 \quad \Leftrightarrow \quad \mathcal{D}_{(\alpha(i}L_{\beta)jkl)} = 0 \quad \Rightarrow \quad w = \frac{9}{2}.$$
 (4.64)

The remaining terms can be solved to find the dimension-5 component of the six form

$$L_{ss\psi\psi\psi\psi} = \frac{i}{48} \epsilon_{\psi\psi\psi\psi}^{ab} (s^i \gamma_{ab})^{\alpha} s^{\beta j} \left(3\mathcal{D}^k_{\alpha} L_{\beta ijk} - \mathcal{D}^k_{\beta} L_{\alpha ijk} \right).$$
(4.65)

The dimension- $\frac{11}{2}$ Bianchi identity is

$$3\mathcal{D}_s L_{ss\psi\psi\psi\psi} - 4\mathcal{D}_{\psi} L_{sss\psi\psi\psi} + 3 \cdot 2iL_{s\gamma(s,s)\psi\psi\psi\psi} - 3 \cdot 4T_{s\psi}{}^{\alpha i}L_{\alpha iss\psi\psi\psi} = 0.$$
(4.66)

The constraints following from the pure spinor projection at this, and at any other level in the Bianchi identities, can be obtained by hitting (4.64) with derivatives.²⁰ We solve this Bianchi identity for the next component to find

$$L_{s\psi\psi\psi\psi\psi\psi} = \frac{1}{192} \epsilon_{\psi\psi\psi\psi\psi}{}^{a} s^{\alpha i} \left(3 \mathcal{D}_{a}^{2jk} + 164i C_{a}{}^{jk} \right) L_{\alpha ijk} - \frac{1}{192} \epsilon_{\psi\psi\psi\psi\psi}{}^{a} (s^{i}\gamma_{ab})^{\alpha} \left(\mathcal{D}^{2b\,jk} + 44i C^{b\,jk} \right) L_{\alpha ijk}.$$
(4.67)

¹⁸Note that this field is of the form of the gauge parameter superfield for supergravity (cf. ref. [43]). Similar observations were made concerning the appearance of superconformal gauge parameter multiplets in the de Rham cohomology in five dimensions in [7, 8] and even in F-theory superspace in [51].

¹⁹Note the similarity of this expression with the pure spinor 0-mode normalization $\langle \lambda^3 \theta^5 \rangle = 1$ [52].

²⁰This is because (4.64) does not project out any irreducible component of Q acting on L. Note that this is in contrast to (4.38) which projects out the γ -trace of $\mathcal{D}_{\gamma(k}G_{a\,ij})$.

The dimension-6 Bianchi

$$2\mathcal{D}_{s}L_{s\psi\psi\psi\psi\psi} + 5\mathcal{D}_{\psi}L_{ss\psi\psi\psi\psi} - 2iL_{\gamma(s,s)\psi\psi\psi\psi\psi} + 2\cdot 5T_{s\psi}^{\alpha i}L_{\alpha i s\psi\psi\psi\psi} - \frac{1}{2}\cdot 5\cdot 4T_{\psi\psi}^{\alpha i}L_{\alpha i ss\psi\psi\psi} = 0, \qquad (4.68)$$

can be solved for the top component of the 6-form to give

$$L_{\psi\psi\psi\psi\psi\psi\psi} = \epsilon_{\psi\psi\psi\psi\psi\psi\psi} \left[\frac{i}{192} \left(\mathcal{D}^{3\alpha ijk} - 12i\mathcal{C}^{\alpha ijk} \right) L_{\alpha ijk} - \frac{1}{9} C_a^{ij} (\mathcal{D}^k \tilde{\gamma}^a L_{ijk}) \right], \qquad (4.69)$$

where $\mathcal{D}_{ijk}^{3\alpha} := \frac{1}{4!} \varepsilon^{\alpha\beta\gamma\delta} \{ \mathcal{D}_{\delta(i)}, [\mathcal{D}_{\gamma j}, \mathcal{D}_{\beta k)}] \}$. The dimension- $\frac{13}{2}$ Bianchi

$$\mathcal{D}_s L_{\psi\psi\psi\psi\psi\psi\psi} - 6\mathcal{D}_{\psi}L_{s\psi\psi\psi\psi\psi\psi} - 6T_{s\psi}{}^{\alpha i}L_{\alpha i\psi\psi\psi\psi\psi} - \frac{1}{2} \cdot 6 \cdot 5T_{\psi\psi}{}^{\alpha i}L_{\alpha is\psi\psi\psi\psi} = 0.$$
(4.70)

does not define any new components and serves only to define the θ^4 -terms in L in terms of derivatives acting on its lower components. Similarly, the dimension-7 Bianchi

$$7\mathcal{D}_{\psi}L_{\psi\psi\psi\psi\psi\psi\psi} - \frac{1}{2} \cdot 7 \cdot 6T_{\psi\psi}{}^{\alpha i}L_{\alpha i\psi\psi\psi\psi\psi\psi} = 0$$
(4.71)

provides the bosonic closure condition (up to torsion) for the six form.

We can solve the condition (4.64) analogously to what was done in section 4.1 by taking

$$L_{\alpha ijk} = \mathcal{D}_{\alpha(i}L_{jk)} + \mathcal{D}_{\alpha}^{l}L_{ijkl} \tag{4.72}$$

and plugging it back in. Note that the first term is Q-exact and is therefore not constrained by (4.64). However, precisely analogously to the case of the constrained prepotential U_{ij} of the gauge field (4.4), the field L_{ijkl} must satisfy the condition

$$\mathcal{D}^m \tilde{\gamma}_{abc} \mathcal{D}_m L_{ijkl} - 384i N_{abc}^{(-)} L_{ijkl} = 0 \tag{4.73}$$

for the constraint (4.64) to hold. In flat space, then, the analogue of Mezinçescu's unconstrained prepotential u^{ij} for the 6-form would be an unconstrained, dimension-2 scalar field ℓ such that $L_{ijkl} = D^4_{ijkl} \ell$.

5 Composite forms

In the previous section, we constructed the de Rham complex of differential forms by sequentially obstructing the closure condition with a form of degree 1 higher. In this section, we investigate the alternative method of building higher-degree forms by wedging forms of lower degree. Analogously to how solving the seemingly trivial closure conditions $d\omega = 0$ resulted in the elucidation of the superspace representations of superconformally covariant *p*-form fields and their coupling to gravity, here we will similarly gain insight into the structure of interactions in superconformal N = (1,0) models and their Lagrangians. In the process, we will derive relations between certain types of composite forms that we compare in section 6.2 to explicit formulæ appearing in the non-abelian tensor hierarchy. We will refer to the forms obtained by wedging lower-degree forms as *composite* forms to distinguish them from the forms above. To minimize additional notation, we will use the same letters in bold font to denote the composite forms. Consider the the composite *p*-form $\omega_p = \omega_q \wedge \omega_{p-q}$. For simplicity of exposition, we mostly focus on the product of only two forms. Then

$$\boldsymbol{\omega}_{s_1\dots s_s\psi_1\dots\psi_{p-s}} = \sum_{r+t=s} c_{rt}^s \,\omega_{s_1\dots s_r\psi_1\dots\psi_{q-r}}\omega_{s_1\dots s_t\psi_1\psi_{p-q-t}} \tag{5.1}$$

for some rational coefficients c_{rt}^s . These are computed by first counting inequivalent permutations of indices and then normalizing the result to 1. For example, the $\mathbf{G}_{ss\psi\psi}$ component of $\mathbf{G} = F \wedge F$ is gotten by writing down the terms $F_{ss}F_{\psi\psi}$ and $F_{s\psi}F_{s\psi}$ and realizing that there are two inequivalent configurations of the indices on the second term, namely $F_{\underline{\alpha}a}F_{\underline{\beta}b}$ and $F_{\underline{\beta}a}F_{\underline{\alpha}b}$ whereas on the first $F_{\underline{\alpha}\beta}$ is equal to $F_{\underline{\beta}\alpha}$ and similarly for F_{ab} . Therefore these 3 terms are weighted as $\frac{1}{3}F_{ss}F_{\psi\psi}$ and $\frac{2}{3}F_{s\psi}F_{s\psi}$. They get a relative sign from the odd permutation $ss\psi\psi \rightarrow s\psis\psi$. Finally, $F_{ss} = 0$ so $\mathbf{G}_{ss\psi\psi} = -\frac{2}{3}F_{s\psi}F_{s\psi}$.

5.1 The composite *p*-form with p = 2 and 3

In the abelian limit $\mathbf{F} := A \wedge A \equiv 0$ so that a single 1-form does not generate a composite 2-form. Given a collection of such forms and a bilinear, skew-symmetric map f, however, one can construct $\mathbf{F}_{ss} := \mathbf{f}(A_s, A_s)$ and its higher components. If, in addition, f maps back into the collection of forms, we can use this composite 2-form as a deformation of the collection of abelian field strengths dA. If one further requires that these maps satisfy the Jacobi identity $\mathbf{f}(\mathbf{f}(A_s, A_s), A_s) = 0$ then this component can be absorbed into a connection $\nabla = D + \mathbf{f}(A, \cdot)$ and we recover the usual formulation of the non-abelian gauge field strength.

The condition $F_{ss} = 0$ is equivalent to $\nabla_s^2 = i \nabla_{\gamma(s,s)}$, defining the vector connection in terms of the spinor connection. With this, the first Bianchi identity becomes equivalent to the associativity of the spinor connection: $0 = \nabla_s (\nabla_s \nabla_s) - (\nabla_s \nabla_s) \nabla_s = i [\nabla_s, \nabla_{\gamma(s,s)}] = i F_{s\gamma(s,s)}$. The rest of the analysis proceeds as in section 4.2.

A composite 3-form is easily constructed as $\mathbf{H} = A \wedge F^{21}$. The properly normalized components

$$\mathbf{H}_{sss} = A_s F_{ss} = 0$$

$$\mathbf{H}_{ss\psi} = \frac{2}{3} A_s F_{s\psi} + \frac{1}{3} A_{\psi} F_{ss} = \frac{2}{3} A_s F_{s\psi}$$

$$\mathbf{H}_{s\psi\psi} = \frac{1}{3} A_s F_{\psi\psi} - \frac{2}{3} A_{\psi} F_{s\psi}$$

$$\mathbf{H}_{dydyb} = A_{\psi} F_{dydb}$$
(5.2)

satisfy the Bianchi identities provided A and F satisfy theirs. That is, **H** is closed provided both A and F are. When dF = 0 but $dA \neq 0$, a short calculation

$$2(d\mathbf{H})_{ss\psi\psi} = 2D_s\mathbf{H}_{s\psi\psi} + 2\partial_{\psi}\mathbf{H}_{ss\psi} - 2i\mathbf{H}_{\gamma(s,s)\psi\psi}$$

 $^{^{21}}$ In the non-abelian case, this can be extended to the full Chern-Simons 3-form. For simplicity of exposition, we work in the abelian limit but allow A and F to be independent fields.

$$= \frac{2}{3} D_{s} A_{s} F_{\psi\psi} - \frac{2}{3} A_{s} D_{s} F_{\psi\psi} - \frac{4}{3} D_{s} A_{\psi} F_{s\psi} - \frac{4}{3} A_{\psi} D_{s} F_{s\psi} + \frac{4}{3} \partial_{\psi} A_{s} F_{s\psi} + \frac{4}{3} A_{s} \partial_{\psi} F_{s\psi} - \frac{2i}{3} A_{\gamma(s,s)} F_{\psi\psi} + \frac{4i}{3} A_{\psi} F_{\gamma(s,s)\psi} = -\frac{2}{3} A_{s} \left(D_{s} F_{\psi\psi} - 2 \partial_{\psi} F_{s\psi} \right) - \frac{4}{3} \left(D_{s} A_{\psi} - \partial_{\psi} A_{s} \right) F_{s\psi} - \frac{2}{3} A_{\psi} \left(D_{s} F_{s\psi} - 2i F_{\gamma(s,s)\psi} \right) = -\frac{4}{3} (dA)_{s\psi} F_{s\psi}$$
(5.3)

shows that $(d\mathbf{H})_{ss\psi\psi} = \mathbf{G}_{ss\psi\psi}$ where $\mathbf{G} = dA \wedge F$.

On the pure spinor subspace, this form is represented simply by the abelian Chern-Simons super-3-form field $\mathbf{C} = AW$. This is a composite analogue of the alternative description of the exact 4-form obstruction (4.47). We will use this form in section 6.2 to obstruct the defining condition (4.33) of the gauge 2-form potential V as

$$QV = \alpha \operatorname{tr}(AW) \tag{5.4}$$

for some parameter α . This equation was proposed in flat space in the form $D_{(i}\tilde{\gamma}_{ab}V_{j)} = \alpha A_{(i}\tilde{\gamma}_{ab}W_{j)}$ in reference [48]. There, it was explained that this deformation is consistent since both sides obey the constraint $D_{(\gamma}^{(i}\sigma^{jk)}{}_{\beta}^{\alpha}$ – traces = 0 where σ stands for the DV and AW combinations on the left-hand side and the right-hand side, respectively. In the pure spinor notation, this observation reduces to the fact that $Q^2 = 0$ on V and that Q(AW) = (QA)W + AQW = 0 by the defining equations (4.2) and (4.12).

5.2 The composite *p*-form with p = 4, 5, and 6

Let $Z^{\alpha i}$ denote a positive chirality Weyl tensor of weight $\frac{3}{2}$ and define its weight-2 field strength

$$\mathbf{\Phi}(Z) := \mathcal{D}_{\alpha i} Z^{\alpha i}. \tag{5.5}$$

Recall that when Z satisfies the condition QZ = 0 (4.33), its associated field strength Φ satisfies the condition (4.26). Additionally, restricting $\Phi = 0$, implies that Z describes the vector multiplet of section 4.2.

For any two such spinor superfields \tilde{Z} and Z, define the bilinear

$$\mathbf{G}_{aij}(\tilde{Z}, Z) := \tilde{Z}_{(i} \gamma_a Z_{j)}. \tag{5.6}$$

When \tilde{Z}, Z satisfy the condition (4.33), as we will henceforth assume, this bilinear satisfies (4.38) and defines a composite version of the 4-form of section 4.4. To see this, let $\mathbf{B}_{ab}(Z) := \mathcal{D}_i \tilde{\gamma}_{ab} Z^i$ denote the 2-form superfield associated to Z in analogy to the definition of the fundamental 2-form (4.35). The composite 4-form has as its lowest non-trivial component $\mathbf{G}_{ss\psi\psi} = -\frac{2}{3} \tilde{\mathbf{B}}_{s\psi} \mathbf{B}_{s\psi}$. In the pure spinor projection this component becomes proportional to $(\lambda \gamma_{\psi\psi a} \lambda) v^i v^j \mathbf{G}_{ij}^a$.

In section 4.4 we found that the top component of the 4-form is given by (4.44). Consider the composite version in the flat limit

$$(*\mathbf{G})_{ab} := D^2_{[aij} \mathbf{G}_{b]}{}^{ij}, \tag{5.7}$$

the bosonic part of which evaluates to

$$\mathbf{G}|_{\text{bose}} = -\frac{3}{16} \left[\tilde{\mathbf{\Phi}} * \mathbf{B} + \mathbf{\Phi} * \tilde{\mathbf{B}} + 6\tilde{\mathbf{B}} \wedge \mathbf{B} \right].$$
(5.8)

In general, this form is not closed. Indeed, straightforward D-algebra gives

$$\partial^{b}(*\mathbf{G})_{ab} = \frac{1}{8}D_{aij}^{2}\left(\mathbf{K}^{ij} + \tilde{\mathbf{K}}^{ij}\right)$$
(5.9)

where we have defined the bilinear $\mathbf{K}_{ij} := iD_{\alpha(i}\tilde{\Phi}Z_{j)}^{\alpha} + \frac{i}{4}\tilde{\Phi}D_{\alpha(i}Z_{j)}^{\alpha}$ and the $\tilde{\mathbf{K}}$ that follows from switching $\tilde{Z} \leftrightarrow Z$. This combination, or its curved version

$$\mathbf{K}_{ij} = i\mathcal{D}_{\alpha(i}\tilde{\mathbf{\Phi}}Z_{j)}^{\alpha} + \frac{i}{4}\tilde{\mathbf{\Phi}}\mathcal{D}_{\alpha(i}Z_{j)}^{\alpha}, \qquad (5.10)$$

is a composite analogue of the defining field of the 5-form multiplet of section 4.5. It is analytic (i.e. it satisfies equation (4.52)) because $\tilde{\Phi}$ satisfies (4.26) [27]. Since $\mathbf{K}(\tilde{Z}, Z) \neq$ $\mathbf{K}(Z, \tilde{Z})$ is not symmetric as a function of \tilde{Z} and Z, the divergenceless vector superfield

$$D_{aij}^{2}\mathbf{K}^{ij} = \frac{3}{2}\tilde{\Phi}\overset{\leftrightarrow}{\partial}_{a}\Phi - \frac{3}{2}\partial^{b}\left(\tilde{\Phi}B_{ab}\right) - \frac{3}{4}\tilde{H}_{abc}^{(-)}B^{bc} + \text{fermions}$$
(5.11)

gives rise to two conserved currents when there are at least two tensor fields present. This will be important when we relate our complex to the non-abelian tensor hierarchy in section 6.2.

The composite 4-form superfield (5.6) can be used to obstruct the defining constraint on the fundamental (i.e. not composite) 3-form field strength superfield Φ :

$$\mathcal{D}_{(i}\tilde{\gamma}_a\mathcal{D}_{j)}\Phi + 16iC_{aij}\Phi = \alpha \mathbf{G}_{aij}.$$
(5.12)

where α is a coupling constant. We now turn to the analysis of this deformation in the case where \tilde{Z} and Z are some combination of vector and tensor multiplets.

5.2.1 The composite 4-form

Specializing $\tilde{Z} = Z = W$ to a single vector multiplet, \mathbf{G}_{aij} becomes the usual supercurrent [43, 53].²² In this special case, the composite 4-form (5.8) reduces to $\mathbf{G} \sim F \wedge F$. It is closed (the 4-form Bianchi identities are not obstructed) since $\mathbf{\Phi} \equiv 0$ and, therefore also, $\mathbf{K} \equiv 0$. The construction is off-shell as there are no tensors present to put it on-shell.

If we couple this form to a fundamental tensor, we recover the fact that the obstructed closure condition (5.12) is the superspace analogue of the Green-Schwarz anomaly equation [48]

$$dH = \alpha F \wedge F. \tag{5.13}$$

²²In six dimensions, this current is analytic only on-shell as $\mathcal{D}_{\gamma(k}\mathbf{G}_{ij}^{a})$ is proportional to the derivative of the vector multiplet auxiliary field X_{ij} .

5.2.2 The composite 5-form

In this section, we take $\tilde{Z} = V$ and Z = W to describe a tensor multiplet and a vector multiplet, respectively. The composite \mathbf{G}_{aij} still describes a 4-form (5.8) but now in terms of a gauge 2-form $\tilde{\mathbf{B}} \to B$ and a field strength 2-form $\mathbf{B} \to F$. The associated composite field strength \mathbf{K} is sourced by the analytic vector-tensor multiplet Lagrangian $-i\mathbf{K}_{ij} = \Phi X_{ij} + \mathcal{D}_{(i}\Phi W_{j)}$ (5.10).

This linear multiplet has an interpretation as a composite version of the super-5-form: consider the lowest component of the composite form $\mathbf{K} = F \wedge H$,

$$\mathbf{K}_{ss\psi\psi\psi} = \frac{1}{10} F_{ss} H_{\psi\psi\psi} - \frac{6}{10} F_{s\psi} H_{s\psi\psi} + \frac{3}{10} F_{\psi\psi} H_{ss\psi} \\ = -\frac{3}{5} F_{s\psi} H_{s\psi\psi} + \frac{3}{10} F_{\psi\psi} H_{ss\psi}.$$
(5.14)

In the pure spinor projection, this reduces to

$$\mathbf{K}_{\lambda\lambda\psi\psi\psi} = \frac{2i}{5}\lambda\gamma_{\psi\psi\psi}\lambda\,v^{i}v^{j}\,\mathbf{K}_{ij} + Q\text{-exact},\tag{5.15}$$

that is, the pure spinor projection of the lowest component of the super-5-form is proportional to \mathbf{K}_{ij} up to a *Q*-exact term.²³ The top component of the 5-form $\mathbf{K}_{\psi\psi\psi\psi\psi}$ is given in terms of \mathbf{K}_{ij} in equation (4.57). Here, we will explicitly compute its flat-space dual $*\mathbf{K}_a = \frac{1}{48} D_{aij} \mathbf{K}^{ij}$. The bosonic part evaluates to

$$\mathbf{K}_{a}\Big|_{\text{bose}} = -\frac{1}{8}\partial^{b} \left(\Phi F_{ab}\right) - \frac{1}{16}H^{(-)}_{abc}F^{bc}$$
(5.16)

where we used the off-shell version of the Maxwell equation

$$D_{aij}^2 X^{ij} = 6i\partial^b F_{ab}. ag{5.17}$$

For the fermionic part, we need the equation

$$D_{\alpha i}F_{\beta}{}^{\gamma} = 2i\partial_{\alpha\beta}W_i^{\gamma} + \frac{2}{3}\delta_{\alpha}^{\gamma}D_{\beta}^jX_{ij} - \frac{1}{3}\delta_{\beta}^{\gamma}D_{\alpha}^jX_{ij}$$
(5.18)

and its consequence

$$D_{aij}^2 W^{\gamma j} = 6i\partial_a W_i^{\gamma} - (\tilde{\gamma}_a)^{\gamma \delta} D_{\delta j} X_i^{j}.$$
(5.19)

With this,

$$\mathbf{K}_{a}\Big|_{\text{fermi}} = \frac{1}{8} D_{k} \Phi \overleftrightarrow{\partial}_{a} W^{k} + \frac{1}{8} D_{k} \Phi \tilde{\gamma}_{a} \gamma_{b} \partial^{b} W^{k}$$
(5.20)

where we have used the fact that the tensor multiplet is on shell (4.28). It is also due to this condition that **K** is divergenceless: algebraically,

$$\partial^a \mathbf{K}_a = -\frac{1}{16} \partial^a H^{(-)}_{abc} F^{bc} - \frac{1}{8} D_k \Box \Phi W^k + \frac{1}{8} \partial^a D_k \Phi \tilde{\gamma}_a \gamma_b \partial^b W^k$$

and these terms are all proportional to the equations of motion (4.28), (4.29), (4.32) of the tensor multiplet.

²³This is not surprising since $Q\mathbf{K}_{\lambda\lambda\psi\psi\psi} \propto QF_{\lambda\psi}H_{\lambda\psi\psi} + F_{\lambda\psi}QH_{\lambda\psi\psi} = 0$ by Bianchi identities and the combination \mathbf{K}_{ij} was originally constructed in reference [27] to satisfy specifically this condition.

5.2.3 The composite 6-form

Finally, we consider the case where both \tilde{Z} and Z describe tensor multiplets \tilde{V} and V. The composite 5-form density resulting from this double-tensor can be understood along the lines of the vector-tensor construction of the previous section by replacing $F \to \tilde{B}$. In particular, $\mathbf{K} = \tilde{B} \wedge H$ and the current (5.16), (5.20) gets modified to the form

$$D_{aij}^{2}\mathbf{K}^{ij} \sim \tilde{\Phi}\overset{\leftrightarrow}{\partial}_{a}\Phi + \partial^{b}\left(\Phi\tilde{B}_{ab}\right) + H_{abc}^{(-)}\tilde{B}^{bc} + D^{k}\Phi\tilde{\gamma}_{a}D_{k}\tilde{\Phi} + D_{k}\Phi\overset{\leftrightarrow}{\partial}_{a}V^{k} + D_{k}\Phi\tilde{\gamma}_{a}\gamma_{b}\partial^{b}V^{k}.$$
(5.21)

Note that this composite is not gauge invariant. It also does not generate a gauge invariant 6-form since the current is conserved. Conservation uses the equations of motion of both multiplets and the fact that $H \wedge \tilde{H} \equiv 0$ for any two anti-self-dual forms H and \tilde{H} . In the case $\tilde{\Phi} = \Phi$, the first term on each line vanishes. We will return to this form in section 6.2.

The fact that we do not generate a closed 6-form with this bilinear is the composite analogue of the observation in section 4.6 that there is no obstruction to the closure of the 5-form K_{ij} once it satisfies the defining relation QK = 0 (4.52). Similarly to the analysis of that section, we can nevertheless define such a composite provided we go beyond bilinears and construct the analogue of $F \wedge F \wedge F$:

$$\mathbf{L}_{sss\psi\psi\psi} = \frac{3\cdot 3}{15} F_{ss} F_{s\psi} F_{\psi\psi} - \frac{3!}{15} F_{s\psi} F_{s\psi} F_{s\psi} = -\frac{2}{5} F_{s\psi} F_{s\psi} F_{s\psi}.$$
 (5.22)

(Other possibilities include $B \wedge F \wedge F$ and $A \wedge F \wedge H$.) In the pure spinor projection, this becomes

$$\mathbf{L}_{sss\psi\psi\psi} \mapsto \frac{16i}{5} c^{3\alpha} v^i v^j v^k (W^3)_{\alpha ijk}$$
(5.23)

giving the composite analogue $\mathbf{L}_{\alpha ijk}$ of the closed 6-form field strength of section 4.6. Note that the Weyl weight of this composite is $w = 3 \cdot \frac{3}{2} = \frac{9}{2}$ in agreement with the condition (4.64).

6 Applications

In the previous sections we studied the structure of differential forms in six-dimensional, N = (1,0) superspace. In this section, we present a selection of applications of these results. Topics we have refrained from discussing include applications to covariant superstring compactifications [36, 54, 55] and related superspace gauge theories (e.g. ref. [56]), the new ambi-twistor strings of [57], the construction of superconformal theories with a second, nonlinearly realized supersymmetry [40], superspaces with boundaries [58], and the comparison to interesting recent lower-dimensional results (e.g. ref. [4, 59, 60]). Instead, we restrict our attention to the two applications that most overlap with the results already derived. These are intended only to motivate the use of superforms and do not represent complete analyses.

6.1 Ectoplasm

This work has its origin in failed attempts to construct the density projection formula [61–63] for curved, six-dimensional, N = (1,0) projective superspace action [27]

$$S = \frac{1}{2\pi} \oint_C (v_i dv^i) \int_M d^6 x \int d^8 \theta E \,\Theta^{(-4)} \mathcal{L}^{++}$$
(6.1)

 (\mathcal{L}^{++}) is a weight-4 analytic superfield). Generally, the Bianchi identities are solved in curved superspace for a p-form with p equal to the dimension of the bosonic space-time. By an extension of Noether's argument defining conserved charges from conserved currents, the components of this form can be shown to define a curved supersymmetric invariant extending the flat-space component action [67]. This invariant is, then, a natural candidate for the component action in curved superspace.²⁴ Six-dimensional, N = (1, 0) superspace has the peculiar property of disallowing the straightforward construction of a natural 6-form. The naïve generalization of two known approaches immediately fails for trivial reasons. One of these extends the observation that it is sometimes possible to construct the top form by wedging two middle-dimensional forms [68]. Applied to six dimensions, we expect to obtain the top form corresponding to the projective measure defined in [27] from the wedge of the 3-form with itself. This fails, however, since the 3-form is self-dual so that the associated 6-form vanishes identically in the flat limit. In curved superspace, it fails to produce the \mathcal{D}^4 part of the analytic measure. An attempt to construct a 6-form from other composites (e.g. three 2-forms) does not generate a forth-order operator acting on a scalar Lagrangian and, therefore, also does not represent the curved analytic measure.

A second attempt to guess the 6-form directly may be made by using Berkovits' Ansatz for the structure of the lowest component of the top form [69]. The proposed component is of the form $L_{\underline{\alpha}\underline{\beta}\underline{\gamma}abc} \sim (\gamma_{abc})_{(\underline{\alpha}\underline{\beta}}f_{\underline{\gamma}})$ with $D_{(\underline{\alpha}}f_{\underline{\beta}}) = 0$. However, this component is pure gauge when interpreted as a Weyl tensor superfield as described in section 4.1. In flat superspace, the Berkovits conjecture can be modified in the Biswas-Siegel approach to pforms in harmonic superspace [3] by constructing a 7-form with one leg in the harmonic sphere $\mathbb{C}P^1$: $\mathcal{L}_{\pm\alpha-\beta-\gamma-abc} = (\gamma_{abc})_{(\alpha\beta}D_{\gamma)}^{-}\mathcal{L}^{++}$. The superfield \mathcal{L}^{++} is required to be analytic $D_{\alpha}^{+}\mathcal{L}^{++}$ and the top component $*K \propto D^{-4}\mathcal{L}^{++}$ reproduces the flat limit of the projective measure of [27]. Somewhat surprisingly, however, the curved superspace Bianchi identities cannot be satisfied for this choice of 6-form: the non-trivial isospin structure of \mathcal{L}^{++} forces the dimension-1 torsions to vanish. (For example, already the first Bianchi identity implies $0 = \mathcal{D}_{(\alpha}^{+}\mathcal{D}_{\beta}^{-}\mathcal{L}^{++} \propto (\gamma^{abc})_{\alpha\beta}N_{abc}\mathcal{L}^{++}$.)

While it is beyond the scope of this work to investigate the question of integration in projective/harmonic superspace in any depth, the application of our results on differential forms already suggests some preliminary insights. For example, the analysis of section 4.5 suggests that the action for a linear multiplet in curved superspace reduces to the component result

$$\int_{N} \mathrm{d}^{5}x \, e \, n^{a} \left\{ \left(\mathcal{D}_{a\,ij}^{2} - 48iC_{a\,ij} \right) K^{ij} + \frac{4i}{5} \Psi_{i}^{b} \tilde{\gamma}_{ab} \mathcal{D}_{j} K^{ij} + \frac{6}{5} \Psi_{i}^{b} \tilde{\gamma}_{abc} \Psi_{j}^{c} K^{ij} \right\}$$
(6.2)

 $^{^{24}}$ For more recent investigations into the general theory of integration on supermanifolds, we refer the reader to [64–66].

with the integral taken over some 5-dimensional bosonic subspace N of M with normal vector n. The full analysis of the four-dimensional analogue of this was carried out in reference [6].

In section 4.6, we found that the correct Ansatz for the 6-form was $L_{sss\psi\psi\psi} = s^i \gamma_{\psi\psi\psi} s^j s^{\gamma k} L_{\alpha ijk}$. Together with the other components derived in that section, we can write down a supersymmetric invariant that, schematically, is given by

$$\int d^6 x \, e \left\{ \mathcal{D}^3 L + \Psi \mathcal{D}^2 L + \Psi \Psi \mathcal{D} L + \Psi \Psi \Psi L \right\}. \tag{6.3}$$

If one further solves the constraint (4.64) on the dimension- $\frac{9}{2}$ component as in (4.72), one obtains a formula for covariantizing the component $\mathcal{D}_{ijkl}^4 L^{ijkl}$. The method used in [31, 70] to obtain the analogous density projection formula starts with precisely such a term and successively constructs the higher components in the gravitino expansion in a Noether-type procedure based on the invariance under projective SL(2; **C**) transformations of the projective superspace action (6.1). Therefore, if the component result from ectoplasm can be checked to be SL(2; **C**)-invariant, it should correspond to the density projection formula for the projective superspace action.

6.2 Abelian tensor hierarchy

The non-abelian tensor hierarchy [38–40] is an attempt to construct a non-abelian gauge theory of forms of degree p > 1 by obstructing the closure of the standard Yang-Mills field strength.²⁵ As we review in appendix B, one introduces a collection of *p*-form potentials $(B^{I}, C_{r}, D_{\alpha}, E_{\mu})$ for p = 2, 3, 4, 5, respectively, extending the standard Yang-Mills potential A^{r} . A collection of linear maps $(\mathbf{h}_{I}^{r}, \mathbf{g}^{rI}, \mathbf{k}_{r}^{\alpha})$ is introduced to obstruct the closure of the *p*form field strength with a (p+1)-form potential. Consistency of this deformation in the nonabelian case requires the extension of the Yang-Mills structure constants \mathbf{f}_{rs}^{t} by a collection of constants denoted by $(\mathbf{d}_{rs}^{I}, \mathbf{b}_{Irs}, \mathbf{c}_{\alpha IJ}, \mathbf{c}_{\alpha r}^{\prime s})$. Finally, superpartners are introduced and the whole model is shown to be superconformally invariant.

The first step in this program is the obstruction of the Bianchi identity of a p-form field strength with a (p + 1)-form field strength. This is precisely the procedure carried out in section 4 to derive the complex of differential forms. Thus, the linearized part of the nonabelian tensor hierarchy is just the construction of this complex. Consider, for example, the case of the vector multiplet field strength W. Shifting $W \to W + h(V)$ obstructs the Bianchi identity (4.15) by the term (4.34). As explained in section 4.3.1, this is the superfield defining the 3-form field strength of section 4.3. It satisfies the condition (4.26) which can, in turn, be obstructed by $\mathbf{g}(G)$ using the 4-form field of section 4.4.

This interpretation of the differential complex refers only to the linearized part of the hierarchy. We see from equation (5.12), however, that certain non-linear parts are captured by introducing the composite deformations from section 5 alongside the fundamental ones. Indeed, if, after constructing the non-abelian tensor hierarchy, one takes the abelian limit $f \rightarrow 0$, one is apparently left with a non-linear theory.²⁶ It seems to be the case, then, that

 $^{^{25}}$ See also [71, 72] for details on the tensor hierarchy algebra and applications and [73] for extensions to four dimensions.

 $^{^{26}}$ There is a subtlety concerning the non-triviality of this limit that we address in footnote 30.

the non-linear but abelian part of the hierarchy is precisely the entire differential complex augmented with composite obstructions. In this sense, one may think of the non-abelian tensor hierarchy as a non-abelian deformation of this complex or "non-abelian ectoplasm" in the terminology of section 6.1. Such structures should be closely related to the gravitational tensor hierarchies of references [74, 75] appearing in gauged supergravity. For recent applications to self-dual non-gravitational theories and their actions, see also [76–79]. Although demonstration of the complete equivalence of the two sides and the non-abelian extension of them is beyond the scope of this paper, some non-trivial comparisons can be made with the results already worked out. The obstructed closure condition (5.13) is central to the construction of the non-abelian tensor hierarchy in which it appears in the form

$$d\mathcal{H} = \mathsf{d}(\mathcal{F}, \mathcal{F}) + \mathsf{g}(\mathcal{G}). \tag{6.4}$$

Here,

- d is the symmetric bi-linear form (extended to act by wedge product on forms) on the space of vector multiplets valued in the space of tensor multiplets,
- $\mathcal{F} = F + h(B)$ is a deformation of the non-abelian 2-form field strength F = dA by a gauge 2-form B,
- $\mathcal{H} = H + g(C)$, is a deformation of the 3-form field strength H = dB by a gauge 3-form C, and
- $\mathcal{G} = G + \mathsf{k}(D)$, is a deformation of the 4-form field strength G = dC by a gauge 4-form D although this term does not enter into the hierarchy at this level since $\mathsf{g} \circ \mathsf{k} = 0$.

As we have seen in section 4.4 and 5.2.1, this condition results from sourcing the defining equation of the tensor superfield strength (5.12). Therefore, provided we shift the pure spinor superfields $W \to W = W + h(V)$, we can capture the b, c, c' = 0 sector of the hierarchy in curved superspace with the constraint

$$\left(\mathcal{D}_{(i}\tilde{\gamma}_{a}\mathcal{D}_{j)} + 16iC_{aij}\right)\Phi = \mathsf{d}(\mathcal{W}_{(i}\gamma_{a}\mathcal{W}_{j)}) + \mathsf{g}(G_{aij}) \tag{6.5}$$

defining the deformed 3-form \mathcal{H} .

It was shown in reference [77] that the *p*-form field strengths with $p \ge 4$ are all *composite*. Therefore, at least when formulated in terms of field strengths, it is possible that this constraint already encodes the entire abelian hierarchy. In fact, the composite "current" (5.6) has a natural extension by the associative *-product of appendix B to $\mathbf{G}_{aij} \sim \mathbf{b}(W, V) + \mathbf{k} \circ \mathbf{c}(V, V)$. Associated to this field is a composite linear superfield \mathbf{K}_{ij} (5.10). An important set of constraints (cf. eq. (3.7) of ref. [38]) in the non-abelian tensor hierarchy is given in superspace by setting

$$\mathbf{b}(\mathbf{K}_{ij}) = 0. \tag{6.6}$$

In the context of section 5.2, this equation implies that the 4-form Bianchi identities are satisfied when the composite obstruction is mapped to the space of 4-forms by b. Equations (5.16) and (5.20) then imply the deformed closure condition (cf. eq. (3.39) and (3.43)

of reference [77])

$$d\mathbf{G} = \mathsf{b}(F, H) + \mathsf{k}(\mathbf{K}),\tag{6.7}$$

where **K** stands for the terms given in equation (5.21). These terms make up the composite 5-form of the tensor hierarchy (cf. eq. (3.43) and (3.49) of reference [77]) in the $c' \rightarrow 0$ limit.²⁷ Thus, we have found that this level of the hierarchy is compactly described by equation (6.5).

By (a deformation of) the discussion in section 4.3.1, the condition (5.13) is equivalent to

$$QV = \mathsf{d}(A, W),\tag{6.8}$$

provided we describe the tensor Φ in terms of its potential V [48]. Using our shifted fields, we can attempt to write the analogous expression for (6.5) in pure spinor superspace. The naïve guess is $QV = \mathsf{d}(A, W) + \mathsf{g}(C)$ for the 3-form potential C described in section 4.4.1. However, according to [77], the associated field strength $G \sim dC$ is *composite*. In fact, it is precisely the composite appearing because the Bianchi identities of dimension ≥ 3 do not close. Therefore, it may even be that an equation of the form

$$DA + A * A = 0$$
 with $D = Q + \partial$ and $A \in \Omega^{\bullet} \otimes K_{\bullet}$ (6.9)

by itself already describes the entire hierarchy in the abelian limit.²⁸ Work is currently underway to investigate this idea and extend it to the full non-abelian hierarchy. For recent closely-related work on such a construction for the bosonic part of the non-abelian tensor hierarchy, see [80, 81] and references therein.

7 Conclusion

In this work we have augmented the curved six-dimensional, N = (1,0) superspace [27] (reviewed in appendix A) with a supercommutative ring of variables (2.2) and exploited its structure to reformulate the superspace analogue of the de Rham complex of differential forms.²⁹ The augmented complex has a "pure spinor" subcomplex (cf. section 3) that is preserved by a second conformal algebra. This construction, roughly speaking, can be interpreted as extending the six-dimensional projective superspace to a super-CR manifold the Lagrangian distribution of which is generated by (3.5). On this subspace the curved superspace spinor derivative reduces to a differential (3.6). Tensor superfields on this subspace that are in the cohomology of this differential define large classes of superconformal primary superfields. They are classified by their Dynkin labels, in terms of which one trivially computes their scaling weights.

 $^{^{27}}$ As we have restricted ourselves to quadratic bilinears in section 5.2, we will not generate the c' contributions of the full tensor hierarchy here.

²⁸It may be of interest to recall here that the complex of *p*-forms was determined uniquely by applying the obstruction procedure to the condition $F_{ss} = 0$. At the beginning of section 4.2, it was explained that this condition is equivalent to the formula QA = 0.

²⁹The flat superspace version of this was recognized in [8] as (a "free resolution" of) the Chevalley-Eilenberg complex of the super-translation algebra with values in superfields.

Certain such superconformal primary superfields are the defining superfields from which de Rham forms are constructed in section 4. Roughly, each of these is the lowestweight member of its respective family. Some forms have alternative representations which also fit naturally into the classification. Composite forms are gotten by multiplying the fundamental forms together. On the pure spinor subcomplex this induces the structure of a differential graded superalgebra. Peculiarities of the superfield structure of certain forms are elucidated by this observation (cf. e.g. section 5.2.2 for the geometrical origin of the projective Lagrangian for systems of tensor and vector multiplets in projective superspace [27]).

Finally, we singled out two particular examples for applications of our results, the full investigations of which are beyond the scope of this work. The first of these is an initial attempt at finding the density projection formula for the projective superspace integral which was the original motivation to consider the differential forms investigated in this work. The second application is to the theory of superconformal models in six dimensions with (at least) N = (1,0) supersymmetry. Specifically, we make a superficial attempt marry the DG algebra structure of the pure spinor subcomplex to the algebraic structure of the non-abelian tensor hierarchy [38–40] (reviewed in appendix B).

Throughout this work, we have relied rather heavily on the notational and calculational simplifications afforded us by the use of auxiliary variables. Stripping off the original supercommutative variables reduces the results to ordinary superspace formulæ. Passing first to the pure spinor subspace and then stripping off only the pure spinor variables, one recovers the analogous results in projective superspace. In this sense, the use of the auxiliary variables is purely a convenience. Nevertheless, we have found the additional structure of such augmented superspaces to be very useful in the theoretical and computational analyses of problems in curved superspace. We hope that our presentation was effective in conveying these, what we perceive to be, significant advantages.

Acknowledgments

It is a pleasure to thank Igor Bandos and Daniel Butter for discussions of their work, Brenno Carlini Vallilo for discussions and collaboration relating this work to the ambitwistor string of reference [57], and Jim Gates for encouragement, support, and references. We are especially indebted to Gabriele Tartaglino-Mazzucchelli for vetting this manuscript, his emphasis of subtle points we had not appreciated regarding integration in curved superspace, and his help with references and Robert Wimmer for carefully reading a previous version of this work, detailed discussions clarifying aspects of the non-abelian tensor hierarchy, and many suggestions for improving the presentation. WDL3 also thanks Stephen Randall for discussions and collaboration on closely-related work that illuminated many confusing aspects regarding superspace cohomology not addressed in the first version of this manuscript, and Richard Wentworth for continuing discussions that have lead to a deeper mathematical understanding of the tools developed and exploited herein.

This work was partially supported by the National Science Foundation grants PHY-0652983 and PHY-0354401 and FONDECYT (Chile) grant number 11100425. CA is supported by the UNAB-DCF M.Sc. scholarship. WDL3 is partially supported by the UMCP

Center for String & Particle Theory. AKR acknowledges participation in the 2013 Student Summer Theoretical Physics Research Session. WDL3 is grateful to the Simons Center for Geometry and Physics for hospitality during the XII Simons Workshop.

A Curved six-dimensional superspace

In this appendix, we collect the results on six-dimensional, N = (1, 0) supergravity used in our analysis of superforms in curved space-time. A detailed understanding of this material is not absolutely necessary to follow the discussion in the main text and serves mainly to fix some notation and introduce the supergravity torsion fields. For additional details, see references [27] and [82].

We denote the local coordinates on curved, six-dimensional, N = (1,0) superspace by $(z^M) = (\theta^{\mu i}, x^m)$. The covariant derivative $(\mathcal{D}_A) = (\mathcal{D}_{\alpha i}, \mathcal{D}_a)$ expands out to

$$\mathcal{D}_A = E_A + \Omega_A + \Phi_A \tag{A.1}$$

where

$$E_A = E_A{}^M \partial_M, \quad \Omega_A = \frac{1}{2} \Omega_A{}^{bc} M_{bc}, \quad \Phi_A = \Phi_A{}^{ij} J_{ij} \tag{A.2}$$

are the coframe, spin connection, and SU(2) connection, respectively. The generators of the superalgebra $\mathfrak{spin}(5,1)\oplus\mathfrak{sp}(1)\subset\mathfrak{osp}(6,2|1)$ are defined by their action on the spinors as

$$[M_{ab}, \mathcal{D}_{\gamma k}] = -\frac{1}{2} (\gamma_{ab})_{\gamma}{}^{\delta} \mathcal{D}_{\delta k} \text{ and } [J_{ij}, \mathcal{D}_{\gamma k}] = -\varepsilon_{k(i} \mathcal{D}_{\gamma j)}.$$
(A.3)

The graded commutation relations of the covariant derivatives define torsions, curvatures, and field strengths

$$[\mathcal{D}_A, \mathcal{D}_B] = T_{AB}{}^C \mathcal{D}_C + \frac{1}{2} R_{AB}{}^{cd} M_{cd} + F_{AB}{}^{ij} J_{ij}.$$
(A.4)

We will work with the supergeometry defined by the relations

$$\begin{aligned} \{\mathcal{D}_{\alpha i}, \mathcal{D}_{\beta j}\} &= 2i\varepsilon_{ij}(\gamma^{a})_{\alpha\beta}\mathcal{D}_{a} + 2iC_{a\,ij}(\gamma^{abc})_{\alpha\beta}M_{bc} + 4i\varepsilon_{ij}N^{abc}(\gamma_{a})_{\alpha\beta}M_{bc} \\ &- 6i\varepsilon_{ij}C_{a}^{kl}(\gamma^{a})_{\alpha\beta}J_{kl} - \frac{8i}{3}N^{abc}(\gamma_{abc})_{\alpha\beta}J_{ij} \\ [\mathcal{D}_{\gamma k}, \mathcal{D}_{a}] &= C_{k}^{bl}(\gamma_{ab})_{\gamma}{}^{\delta}\mathcal{D}_{\delta l} + N_{abc}(\gamma^{bc})_{\gamma}{}^{\delta}\mathcal{D}_{\delta k} + \frac{1}{2}R_{\gamma ka}{}^{bc}M_{bc} \\ &+ \left((\gamma_{a})_{\gamma\delta}\mathcal{C}_{k}^{\delta\,ij} - 6\delta_{k}^{i}\mathcal{C}_{a\,\gamma}{}^{j} + 5\delta_{k}^{i}(\gamma_{a})_{\gamma\delta}\left[\mathcal{C}^{\delta j} - \frac{1}{3}\mathcal{N}^{\delta j}\right]\right)J_{ij}. \end{aligned}$$
(A.5)

The curvature term of dimension- $\frac{3}{2}$ is an unilluminating function of the dimension- $\frac{3}{2}$ torsion so we do not reproduce it here. The dimension- $\frac{3}{2}$ torsion components C and N appear in the higher components of the *p*-forms. Their definitions are as the irreducible components

$$\mathcal{D}_{\gamma k} C_{a i j} = \mathcal{C}_{a \gamma k i j} + (\gamma_{a})_{\gamma \delta} \mathcal{C}^{\delta}_{i j k} + \varepsilon_{k (i} \mathcal{C}_{a \gamma j)} + \varepsilon_{k (i} (\gamma_{a})_{\gamma \delta} \mathcal{C}^{\delta}_{j)}$$

$$\mathcal{D}_{\gamma k} N_{\alpha \beta} = \mathcal{N}_{\gamma k \alpha \beta} + \check{\mathcal{N}}_{\gamma k \alpha \beta}$$

$$\mathcal{D}_{\gamma k} N^{\alpha \beta} = \mathcal{N}_{\gamma k}{}^{\alpha \beta} + \delta^{(\alpha}_{\gamma} \mathcal{N}^{\beta)}_{k}.$$
 (A.6)

These components are constrained by the supergravity Bianchi identities to be [27, 82]

$$\begin{aligned}
\mathcal{C}_{a\,\gamma k\,ij} &= 0 & \mathcal{N}_{\gamma k\,\alpha\beta} = 0 \\
\mathcal{C}_{ijk}^{\delta} &= -\frac{1}{6} (\tilde{\gamma}^b)^{\delta\beta} \mathcal{D}_{\beta(k} C_{b\,ij)} & \check{\mathcal{N}}_{\gamma k\,\alpha\beta} = -\frac{3}{4} (\gamma^a)_{\gamma(\alpha} \mathcal{C}_{a\,\beta)k} \\
\mathcal{C}_{a\,\beta j} &= \frac{2}{3} \Pi_{a\,\beta}^{c\,\gamma} \mathcal{D}_{\gamma}^i C_{a\,ij} & \mathcal{N}_{\gamma k}^{\alpha\beta} = \mathcal{D}_{\gamma k} N^{\alpha\beta} - \frac{2}{5} \delta_{\gamma}^{(\alpha} \mathcal{D}_{\delta k} N^{\beta)\delta} \\
\mathcal{C}^{\gamma k} &= -\frac{1}{9} \mathcal{D}_{\delta l} C^{\delta\gamma\,lk} & \mathcal{N}^{\alpha i} = \frac{2}{5} \mathcal{D}_{\beta}^i N^{\beta\alpha},
\end{aligned} \tag{A.7}$$

where

$$\Pi_{a\alpha}^{b\beta} = \delta_a^b \delta_\alpha^\beta + \frac{1}{6} (\gamma_a \tilde{\gamma}^b)_\alpha^\beta \tag{A.8}$$

is the projector onto the γ -traceless subspace of the spinor-vector representation: $\Pi_{a\alpha}^{b\beta}(\gamma_b)_{\beta\gamma} \equiv 0$ and $(\gamma^a)^{\gamma\alpha} \Pi_{a\alpha}^{b\beta} \equiv 0$.

Super-Weyl transformations preserving this geometry are generated by a real, unconstrained, scalar superfield $\sigma(z)$. The transformations that preserve the covariant derivative algebra act as

$$\delta \mathcal{D}_{\alpha i} = \frac{1}{2} \sigma \mathcal{D}_{\alpha i} - 2(\mathcal{D}_{\beta i} \sigma) M_{\alpha}{}^{\beta} + 4(\mathcal{D}_{\alpha}{}^{j} \sigma) J_{ij}$$
(A.9)

$$\delta \mathcal{D}_a = \sigma \mathcal{D}_a - \frac{i}{2} (\mathcal{D}^k \sigma) \tilde{\gamma}_a \mathcal{D}_k - (\mathcal{D}^b \sigma) M_{ab} - \frac{i}{8} (\mathcal{D}^i \tilde{\gamma}_a \mathcal{D}^j \sigma) J_{ij}$$
(A.10)

on the covariant derivatives and as

$$\delta C_{a\,ij} = \sigma C_{a\,ij} + \frac{i}{8} (\mathcal{D}_{(i} \tilde{\gamma}_c \mathcal{D}_{j)} \sigma) \tag{A.11}$$

$$\delta N_{abc} = \sigma N_{abc} - \frac{i}{32} (\mathcal{D}^k \tilde{\gamma}_{abc} \mathcal{D}_k \sigma)$$
(A.12)

on the dimension-1 torsions.

In section 2 we introduce a commuting spinor $s^{\alpha i}$ that plays the role of $d\theta^{\alpha i}$ in the algebra of exterior superforms. The product of two of such basis elements decomposes into two parts

$$s^{\alpha i}s^{\beta j} = -\frac{1}{8} \varepsilon^{ij}(\tilde{\gamma}_a)^{\alpha\beta}\gamma^a(s,s) + \frac{1}{48}(\tilde{\gamma}^{abc})^{\alpha\beta}\omega^{ij}_{abc}(s,s)$$
(A.13)

where we have defined the vector and self-dual 3-form components

$$\gamma^a(s,s) := s^k \gamma^a s_k \quad \text{and} \quad \omega^{ij}_{abc}(s,s) := s^{(i} \gamma_{abc} s^{j)}. \tag{A.14}$$

The vector $\gamma^a(s,s)$ is null as follows from the general identity

$$(s^i \gamma^a s_i)(s^j \gamma_a \xi_j) \equiv 0 \tag{A.15}$$

which holds for any chiral spinor ξ (because $(\gamma^a)_{\alpha\beta}(\gamma_a)_{\gamma\delta} = 2\varepsilon_{\alpha\beta\gamma\delta}$ whereas the isospin indices range only over 2 values). It is also orthogonal to the triplet of 3-forms $\gamma^a(s,s) \omega^{ij}_{abc}(s,s) \equiv 0$. Many Fierz identities can be derived from these basic relations by polarizing on s (i.e. replacing $s \to s + t + u$, expanding, and collecting like powers). Projecting $s \mapsto \lambda \otimes v$ to the product of a commuting chiral spinor λ^{α} and isotwisor v^i kills the vector part and isolates the self-dual 3-form part of the bilinear. A chiral spinor with the property

$$\lambda^{\alpha}\lambda^{\beta} = \frac{1}{3!2^3} (\tilde{\gamma}^{abc})^{\alpha\beta} \lambda \gamma_{abc} \lambda \tag{A.16}$$

is called *pure* (see e.g. reference [83]) so we will refer to this projection as the pure spinor projection.

The constant, commuting spinor s combines with the covariant derivative to define the odd derivation $\mathcal{D}_s = s^{\alpha i} \mathcal{D}_{\alpha i}$ which squares to

$$\mathcal{D}_{s}^{2} = i\mathcal{D}_{\gamma(s,s)} + 2iN_{\gamma(s,s)ab}M^{ab} - 3iC_{\gamma(s,s)}^{ij}J_{ij} + i\omega_{ij}^{abc}(s,s)\left(C_{c}^{ij}M_{ab} - \frac{4}{3}N_{abc}J^{ij}\right).$$
(A.17)

In section 2, we define the projected derivation $Q = \mathcal{D}_{\lambda \otimes v}$. Its square reduces to

$$Q^{2} = i(\lambda \gamma^{abc} \lambda) v^{i} v^{j} \left(C_{c\,ij} M_{ab} - \frac{4}{3} N_{abc} J_{ij} \right). \tag{A.18}$$

In section 3, we introduce a complex of spaces on which this square vanishes, thereby promoting Q to a differential.

B The non-abelian tensor hierarchy

In section 6.2, we describe an application of the results on the structure of fundamental and composite p-forms to the (abelian part of the) non-abelian tensor hierarchy [38–40]. In this appendix, we review very briefly the underlying cohomological structure of this hierarchy and propose a reformulation of it in terms of a generalized field strength satisfying a Maurer-Cartan equation.

The non-abelian tensor hierarchy is built on a collection of p-form gauge fields $(A^r, B^I, C_r, D_\alpha, E_\mu)$ where p = 1, 2, 3, 4, 5, respectively. The representation indices take values in a collection of vector spaces that fit into a chain complex

$$K_{\bullet} = \dots \longrightarrow K_3 \xrightarrow{\partial} K_2 \xrightarrow{\partial} K_1 \xrightarrow{\partial} K_0 \longrightarrow 0.$$
 (B.1)

Here the representation space of a gauge *p*-form is denoted by K_{p-1} . The first few terms of the differential are denoted by $(\partial_1, \partial_2, \partial_3) = (h, g, k)$.

Introduce the de Rham complex

$$\Omega^{\bullet} = 0 \longrightarrow \Omega^0 \xrightarrow{d} \Omega^1 \xrightarrow{d} \Omega^2 \xrightarrow{d} \Omega^3 \longrightarrow \dots$$
(B.2)

and consider the double complex $\Omega^{\bullet} \otimes K_{\bullet}$ with differential $D = d + \partial$. Let A denote an element of total degree 1 so that A represents the collection $(A^r, B^I, C_r, D_{\alpha}, E_{\mu})$ of bi-degrees (1,0), (2,-1), (3,-2), (4,-3), (5,-4), respectively. Then

$$\mathsf{F} := D\mathsf{A} \tag{B.3}$$

is a form of total degree 2 representing $(F^r, H^I, G_r, K_\alpha, L_\mu) = (dA^r + \mathbf{h}_I^r B^I, dB^I + \mathbf{g}^{Ir} C_r, dC_r + \mathbf{k}_r^\alpha D_\alpha, dD_\alpha + \mathbf{I}_\alpha^\mu E_\mu, dE_\mu)$. Then

$$D\mathsf{F} = 0 \quad \Leftrightarrow \quad \begin{cases} dF^r = \mathsf{h}_I^r H^I \\ dH^I = \mathsf{g}^{Ir} G_r \\ dG_r = \mathsf{k}_r^\alpha K_\alpha \\ dK_\alpha = \mathsf{I}_\alpha^\mu L_\mu \\ dL_\mu = 0. \end{cases}$$
(B.4)

Analogously to the algebra structure given to the de Rham complex by the wedge product, it turns out that K_{\bullet} can be given an algebra structure by defining a collection of maps

$$*: K_p \times K_q \to K_{p+q+1}. \tag{B.5}$$

When we wish to distinguish them, we will denote the non-vanishing restrictions of the product by $(\mathsf{d}_{rs}^I, \mathsf{b}_{Irs}, \mathsf{c}_{\alpha IJ}, \mathsf{c}_{\alpha r}'^s)$ [38]. Together with the tensors defining the differential on K_{\bullet} , these define an action

$$\triangleright : K_0 \times K_{\bullet} \longrightarrow K_{\bullet}$$
$$(X, a) \mapsto \triangleright_X a \tag{B.6}$$

of K_{\bullet} on itself shown in reference [84] to make the *-product associative (a*b)*c = a*(b*c)and the action \triangleright_X a derivation $\triangleright_X(a*b) = (\triangleright_X a)*b + a*(\triangleright_X b)$ of the resulting algebra $\forall a, b, c \in K_{\bullet}$. Here, we will extend this product to the double complex by wedge, that is, we interpret * on the double complex to mean * on K_{\bullet} together with \wedge on Ω^{\bullet} .

We are now in a position to use the *-product to deform the closure condition (B.4) on F. To do this, one should first extend the differential on the double complex to a connection $\nabla = D + \triangleright_A$. With this, one can define the non-abelian field strength $\mathsf{F} := \nabla^2$. Finally, one writes $\nabla \mathsf{F} + \mathsf{F} * \mathsf{F} = 0$.

The non-abelian tensor hierarchy appears to have a non-linear abelian limit obtained by setting $f \rightarrow 0$ after deforming the complex with the *-product.³⁰ For simplicity, we will work in this limit. Then the deformed version of the closure condition (B.4) may be postulated to be the Maurer-Cartan equation

$$D\mathsf{F} + \mathsf{F} * \mathsf{F} = 0. \tag{B.7}$$

This equation expands out to

$$dF^r = \mathsf{h}_I^r H^I \tag{B.8a}$$

$$dH^{I} = \mathsf{g}^{Ir}G_{r} + \mathsf{d}^{I}_{rs}F^{r} \wedge F^{s} \tag{B.8b}$$

³⁰The naïve limit appears to be non-trivial but we have not completed the analysis required to show the existence of non-trivial solutions to all the conditions this limit affects. Irrespective of this, the consistency of the resulting, perhaps formal, structure is a prerequisite to the extension to the full non-abelian hierarchy. The latter has been explicitly checked to have non-trivial solutions [38]. We thank Robert Wimmer for emphasizing these important points to us.

$$dG_r = \mathsf{k}_r^\alpha K_\alpha + \mathsf{b}_{Irs} F^s \wedge H^I \tag{B.8c}$$

$$dK_{\alpha} = \mathsf{I}^{\mu}_{\alpha}L_{\mu} + \mathsf{c}_{\alpha IJ}H^{I} \wedge H^{J} + \mathsf{c}^{\prime s}_{\alpha r}F^{r} \wedge G_{s} \tag{B.8d}$$

$$dL_{\mu} = 0. \tag{B.8e}$$

In section 6.2 we connect this construction to the sourced and composite p-form complices of sections 4 and 5. For recent work in this direction, see [80, 81].

Open Access. This article is distributed under the terms of the Creative Commons Attribution License (CC-BY 4.0), which permits any use, distribution and reproduction in any medium, provided the original author(s) and source are credited.

References

- S.J. Gates, M.T. Grisaru, M. Roček and W. Siegel, Superspace or one thousand and one lessons in supersymmetry, Front. Phys. 58 (1983) 1 [hep-th/0108200] [INSPIRE].
- [2] S.J. Gates, Jr., Super p form gauge superfields, Nucl. Phys. B 184 (1981) 381 [INSPIRE].
- [3] T. Biswas and W. Siegel, N = 2 harmonic superforms, multiplets and actions, JHEP 11 (2001) 004 [hep-th/0105084] [INSPIRE].
- [4] S.M. Kuzenko and J. Novak, On supersymmetric Chern-Simons-type theories in five dimensions, JHEP 02 (2014) 096 [arXiv:1309.6803] [INSPIRE].
- [5] G. Bossard, P.S. Howe and K.S. Stelle, Invariants and divergences in half-maximal supergravity theories, JHEP 07 (2013) 117 [arXiv:1304.7753] [INSPIRE].
- [6] D. Butter, S.M. Kuzenko and J. Novak, The linear multiplet and ectoplasm, JHEP 09 (2012) 131 [arXiv:1205.6981] [INSPIRE].
- [7] S.J. Gates, W.D. Linch and S. Randall, Superforms in five-dimensional, N = 1 superspace, JHEP 05 (2015) 049 [arXiv:1412.4086] [INSPIRE].
- [8] W.D. Linch and S. Randall, Superspace de Rham complex and relative cohomology, JHEP 09 (2015) 190 [arXiv:1412.4686] [INSPIRE].
- [9] G. Bossard, P.S. Howe and K.S. Stelle, Anomalies and divergences in N = 4 supergravity, Phys. Lett. B 719 (2013) 424 [arXiv:1212.0841] [INSPIRE].
- [10] J. Greitz and P.S. Howe, Maximal supergravity in D = 10: forms, Borcherds algebras and superspace cohomology, JHEP 08 (2011) 146 [arXiv:1103.5053] [INSPIRE].
- [11] G. Bossard, P.S. Howe, U. Lindström, K.S. Stelle and L. Wulff, Integral invariants in maximally supersymmetric Yang-Mills theories, JHEP 05 (2011) 021 [arXiv:1012.3142]
 [INSPIRE].
- [12] S.M. Kuzenko and J. Novak, Supergravity-matter actions in three dimensions and Chern-Simons terms, JHEP 05 (2014) 093 [arXiv:1401.2307] [INSPIRE].
- [13] S.M. Kuzenko, J. Novak and G. Tartaglino-Mazzucchelli, N = 6 superconformal gravity in three dimensions from superspace, JHEP 01 (2014) 121 [arXiv:1308.5552] [INSPIRE].
- [14] D. Butter, S.M. Kuzenko, J. Novak and G. Tartaglino-Mazzucchelli, Conformal supergravity in three dimensions: off-shell actions, JHEP 10 (2013) 073 [arXiv:1306.1205] [INSPIRE].

- [15] D. Butter, S.M. Kuzenko, J. Novak and G. Tartaglino-Mazzucchelli, Conformal supergravity in five dimensions: new approach and applications, JHEP 02 (2015) 111 [arXiv:1410.8682] [INSPIRE].
- [16] S.M. Kuzenko, J. Novak and I.B. Samsonov, The anomalous current multiplet in 6D minimal supersymmetry, JHEP 02 (2016) 132 [arXiv:1511.06582] [INSPIRE].
- [17] S.J. Gates, Jr. and H. Nishino, Manifestly supersymmetric $O(\alpha')$ superstring corrections in new D = 10, N = 1 supergravity Yang-Mills theory, Phys. Lett. **B** 173 (1986) 52 [INSPIRE].
- [18] S.J. Gates, Using superspace techniques to construct effective actions for massless string states, in Proceedings, Quarks '86, Tbilisi Georgia (1986), pg. 151 [INSPIRE].
- [19] S.J. Gates, Jr. and S. Vashakidze, On D = 10, N = 1 supersymmetry, superspace geometry and superstring effects, Nucl. Phys. B 291 (1987) 172 [INSPIRE].
- [20] N. Berkovits and P.S. Howe, The cohomology of superspace, pure spinors and invariant integrals, JHEP 06 (2008) 046 [arXiv:0803.3024] [INSPIRE].
- [21] W. Siegel and M. Roček, On off-shell supermultiplets, Phys. Lett. B 105 (1981) 275 [INSPIRE].
- [22] A. Galperin, E. Ivanov, S. Kalitsyn, V. Ogievetsky and E. Sokatchev, Unconstrained N = 2 matter, Yang-Mills and supergravity theories in harmonic superspace, Class. Quant. Grav. 1 (1984) 469 [Erratum ibid. 2 (1985) 127] [INSPIRE].
- [23] A.S. Galperin, E.A. Ivanov, V.I. Ogievetsky and E.S. Sokatchev, *Harmonic superspace*, Cambridge University Press, Cambridge U.K. (2001) [INSPIRE].
- [24] A. Karlhede, U. Lindström and M. Roček, Selfinteracting tensor multiplets in N = 2 superspace, Phys. Lett. B 147 (1984) 297 [INSPIRE].
- [25] U. Lindström and M. Roček, New hyper-Kähler metrics and new supermultiplets, Commun. Math. Phys. 115 (1988) 21 [INSPIRE].
- [26] U. Lindström and M. Roček, N = 2 super Yang-Mills theory in projective superspace, Commun. Math. Phys. 128 (1990) 191 [INSPIRE].
- [27] W.D. Linch, III and G. Tartaglino-Mazzucchelli, Six-dimensional supergravity and projective superfields, JHEP 08 (2012) 075 [arXiv:1204.4195] [INSPIRE].
- [28] S.M. Kuzenko and G. Tartaglino-Mazzucchelli, Five-dimensional N = 1 AdS superspace: geometry, off-shell multiplets and dynamics, Nucl. Phys. B 785 (2007) 34
 [arXiv:0704.1185] [INSPIRE].
- [29] S.M. Kuzenko and G. Tartaglino-Mazzucchelli, Five-dimensional superfield supergravity, Phys. Lett. B 661 (2008) 42 [arXiv:0710.3440] [INSPIRE].
- [30] S.M. Kuzenko and G. Tartaglino-Mazzucchelli, Super-Weyl invariance in 5D supergravity, JHEP 04 (2008) 032 [arXiv:0802.3953] [INSPIRE].
- [31] S.M. Kuzenko and G. Tartaglino-Mazzucchelli, 5D supergravity and projective superspace, JHEP 02 (2008) 004 [arXiv:0712.3102] [INSPIRE].
- [32] S.M. Kuzenko, U. Lindström, M. Roček and G. Tartaglino-Mazzucchelli, 4D N = 2 supergravity and projective superspace, JHEP 09 (2008) 051 [arXiv:0805.4683] [INSPIRE].
- [33] S.M. Kuzenko, U. Lindström, M. Roček and G. Tartaglino-Mazzucchelli, On conformal supergravity and projective superspace, JHEP 08 (2009) 023 [arXiv:0905.0063] [INSPIRE].

- [34] G. Tartaglino-Mazzucchelli, 2D N = (4, 4) superspace supergravity and bi-projective superfields, JHEP **04** (2010) 034 [arXiv:0911.2546] [INSPIRE].
- [35] G. Tartaglino-Mazzucchelli, On 2D N = (4, 4) superspace supergravity, Phys. Part. Nucl. Lett. 8 (2011) 251 [arXiv:0912.5300] [INSPIRE].
- [36] O. Chandía, W.D. Linch, III and B.C. Vallilo, The covariant superstring on K3, arXiv:1109.3200 [INSPIRE].
- [37] S. Randall, The structure of superforms, arXiv:1412.4448 [INSPIRE].
- [38] H. Samtleben, E. Sezgin and R. Wimmer, (1,0) superconformal models in six dimensions, JHEP 12 (2011) 062 [arXiv:1108.4060] [INSPIRE].
- [39] H. Samtleben, E. Sezgin, R. Wimmer and L. Wulff, New superconformal models in six dimensions: gauge group and representation structure, PoS(CORFU2011)071
 [arXiv:1204.0542] [INSPIRE].
- [40] H. Samtleben, E. Sezgin and R. Wimmer, Six-dimensional superconformal couplings of non-Abelian tensor and hypermultiplets, JHEP 03 (2013) 068 [arXiv:1212.5199] [INSPIRE].
- [41] M. Cederwall, B.E.W. Nilsson and D. Tsimpis, Spinorial cohomology and maximally supersymmetric theories, JHEP 02 (2002) 009 [hep-th/0110069] [INSPIRE].
- [42] S.J. Gates, Jr. and W. Siegel, Variant superfield representations, Nucl. Phys. B 187 (1981) 389 [INSPIRE].
- [43] P.S. Howe, G. Sierra and P.K. Townsend, Supersymmetry in six-dimensions, Nucl. Phys. B 221 (1983) 331 [INSPIRE].
- [44] J. Koller, A six-dimensional superspace approach to extended superfields, Nucl. Phys. B 222 (1983) 319 [INSPIRE].
- [45] L. Mezinçescu, On the superfield formulation of O(2) supersymmetry, Russia (1979) [INSPIRE].
- [46] W. Siegel, Superfields in higher dimensional space-time, Phys. Lett. B 80 (1979) 220 [INSPIRE].
- [47] E. Bergshoeff, E. Sezgin and A. Van Proeyen, Superconformal tensor calculus and matter couplings in six-dimensions, Nucl. Phys. B 264 (1986) 653 [Erratum ibid. B 598 (2001) 667]
 [INSPIRE].
- [48] E. Bergshoeff, E. Sezgin and E. Sokatchev, *Couplings of selfdual tensor multiplet in six-dimensions*, *Class. Quant. Grav.* **13** (1996) 2875 [hep-th/9605087] [INSPIRE].
- [49] J.M. Drummond and P.S. Howe, Codimension zero superembeddings, Class. Quant. Grav. 18 (2001) 4477 [hep-th/0103191] [INSPIRE].
- [50] P.S. Howe, K.S. Stelle and P.K. Townsend, Superactions, Nucl. Phys. B 191 (1981) 445 [INSPIRE].
- [51] W.D. Linch and W. Siegel, *F-theory superspace*, arXiv:1501.02761 [INSPIRE].
- [52] N. Berkovits, Super Poincaré covariant quantization of the superstring, JHEP 04 (2000) 018
 [hep-th/0001035] [INSPIRE].
- [53] P.S. Howe, K.S. Stelle and P.K. Townsend, Supercurrents, Nucl. Phys. B 192 (1981) 332 [INSPIRE].

- [54] O. Chandía, W.D. Linch, III and B.C. Vallilo, Compactification of the heterotic pure spinor superstring I, JHEP 10 (2009) 060 [arXiv:0907.2247] [INSPIRE].
- [55] O. Chandía, W.D. Linch and B. Carlini Vallilo, Compactification of the heterotic pure spinor superstring II, JHEP 10 (2011) 098 [arXiv:1108.3555] [INSPIRE].
- [56] M. Cederwall and B.E.W. Nilsson, Pure spinors and D = 6 super-Yang-Mills, arXiv:0801.1428 [INSPIRE].
- [57] L. Mason and D. Skinner, Ambitwistor strings and the scattering equations, JHEP 07 (2014) 048 [arXiv:1311.2564] [INSPIRE].
- [58] P.S. Howe, T.G. Pugh, K.S. Stelle and C. Strickland-Constable, *Ectoplasm with an edge*, *JHEP* 08 (2011) 081 [arXiv:1104.4387] [INSPIRE].
- [59] D. Butter, N = 2 conformal superspace in four dimensions, JHEP **10** (2011) 030 [arXiv:1103.5914] [INSPIRE].
- [60] J. Novak, Superform formulation for vector-tensor multiplets in conformal supergravity, JHEP 09 (2012) 060 [arXiv:1205.6881] [INSPIRE].
- [61] S.J. Gates, Jr., Ectoplasm has no topology: the prelude, hep-th/9709104 [INSPIRE].
- [62] S.J. Gates, Jr., Ectoplasm has no topology, Nucl. Phys. B 541 (1999) 615 [hep-th/9809056]
 [INSPIRE].
- S.J. Gates, Jr., M.T. Grisaru, M.E. Knutt-Wehlau and W. Siegel, Component actions from curved superspace: normal coordinates and ectoplasm, Phys. Lett. B 421 (1998) 203
 [hep-th/9711151] [INSPIRE].
- [64] L. Castellani, R. Catenacci and P.A. Grassi, Supergravity actions with integral forms, Nucl. Phys. B 889 (2014) 419 [arXiv:1409.0192] [INSPIRE].
- [65] L. Castellani, R. Catenacci and P.A. Grassi, The geometry of supermanifolds and new supersymmetric actions, Nucl. Phys. B 899 (2015) 112 [arXiv:1503.07886] [INSPIRE].
- [66] L. Castellani, R. Catenacci and P.A. Grassi, Hodge dualities on supermanifolds, Nucl. Phys. B 899 (2015) 570 [arXiv:1507.01421] [INSPIRE].
- [67] W. Siegel, Fields, hep-th/9912205 [INSPIRE].
- [68] S.J. Gates, Jr., S.M. Kuzenko and G. Tartaglino-Mazzucchelli, Chiral supergravity actions and superforms, Phys. Rev. D 80 (2009) 125015 [arXiv:0909.3918] [INSPIRE].
- [69] N. Berkovits, Explaining pure spinor superspace, hep-th/0612021 [INSPIRE].
- [70] S.M. Kuzenko and G. Tartaglino-Mazzucchelli, Different representations for the action principle in 4D N = 2 supergravity, JHEP 04 (2009) 007 [arXiv:0812.3464] [INSPIRE].
- [71] J. Palmkvist, The tensor hierarchy algebra, J. Math. Phys. 55 (2014) 011701
 [arXiv:1305.0018] [INSPIRE].
- [72] J. Greitz, P. Howe and J. Palmkvist, The tensor hierarchy simplified, Class. Quant. Grav. 31 (2014) 087001 [arXiv:1308.4972] [INSPIRE].
- [73] K. Becker, M. Becker, W.D. Linch and D. Robbins, *Abelian tensor hierarchy in 4D*, N = 1 superspace, *JHEP* **03** (2016) 052 [arXiv:1601.03066] [INSPIRE].
- [74] B. de Wit and H. Samtleben, Gauged maximal supergravities and hierarchies of non-Abelian vector-tensor systems, Fortsch. Phys. 53 (2005) 442 [hep-th/0501243] [INSPIRE].

- [75] B. de Wit, H. Nicolai and H. Samtleben, Gauged supergravities, tensor hierarchies and M-theory, JHEP 02 (2008) 044 [arXiv:0801.1294] [INSPIRE].
- [76] I. Bandos, H. Samtleben and D. Sorokin, Duality-symmetric actions for non-Abelian tensor fields, Phys. Rev. D 88 (2013) 025024 [arXiv:1305.1304] [INSPIRE].
- [77] I.A. Bandos, Non-Abelian tensor hierarchy in (1,0) D = 6 superspace, JHEP 11 (2013) 203 [arXiv:1308.2397] [INSPIRE].
- [78] I.L. Buchbinder and N.G. Pletnev, Construction of 6D supersymmetric field models in N = (1,0) harmonic superspace, Nucl. Phys. B 892 (2015) 21 [arXiv:1411.1848] [INSPIRE].
- [79] I.L. Buchbinder and N.G. Pletnev, Leading low-energy effective action in the 6D hypermultiplet theory on a vector/tensor background, Phys. Lett. B 744 (2015) 125
 [arXiv:1502.03257] [INSPIRE].
- [80] C. Sämann and M. Wolf, Non-Abelian tensor multiplet equations from twistor space, Commun. Math. Phys. 328 (2014) 527 [arXiv:1205.3108] [INSPIRE].
- [81] C. Säemann and M. Wolf, Six-dimensional superconformal field theories from principal 3-bundles over twistor space, Lett. Math. Phys. 104 (2014) 1147 [arXiv:1305.4870]
 [INSPIRE].
- [82] C. Arias, The geometry of 6D, N = (1,0) superspace and its matter couplings, M.Sc. thesis, Universidad Andrés Bello, Santiago Chile March 2014 [arXiv:1511.00335] [INSPIRE].
- [83] N. Berkovits and S.A. Cherkis, Higher-dimensional twistor transforms using pure spinors, JHEP 12 (2004) 049 [hep-th/0409243] [INSPIRE].
- [84] S. Palmer and C. Sämann, Six-dimensional (1,0) superconformal models and higher gauge theory, J. Math. Phys. 54 (2013) 113509 [arXiv:1308.2622] [INSPIRE].